

This item is the archived peer-reviewed author-version of:

Plasma-based CO₂ conversion : how to correctly analyze the performance?

Reference:

Wanten Bart, Vertongen Rani, De Meyer Robin, Bogaerts Annemie.- Plasma-based CO₂ conversion : how to correctly analyze the performance? Journal of energy chemistry - ISSN 2095-4956 - 86(2023), p. 180-196 Full text (Publisher's DOI): https://doi.org/10.1016/J.JECHEM.2023.07.005 To cite this reference: https://hdl.handle.net/10067/1987090151162165141

uantwerpen.be

Institutional repository IRUA

Plasma-based CO₂ conversion: How to correctly analyze the performance?

Bart Wanten^{a,1,*}, Rani Vertongen^{a,1}, Robin De Meyer^{a,b}, Annemie Bogaerts^{a,*}

^a Research Group PLASMANT, Department of Chemistry, University of Antwerp, Universiteitsplein 1, 2610 Antwerp, Belgium.

^b Research group EMAT, Department of Physics, University of Antwerp, Groenenborgerlaan 171, 2020 Antwerp, Belgium

*Corresponding authors.

E-mail addresses: <u>bart.wanten@uantwerpen.be</u> (B. Wanten);

annemie.bogaerts@uantwerpen.be (A. Bogaerts)

¹These authors contributed equally to this work.

Abstract

Plasma-based CO_2 conversion is promising for carbon capture and utilization. However, inconsistent reporting of the performance metrics makes it difficult to compare plasma processes systematically, complicates elucidating the underlying mechanisms and compromises further development of this technology. Therefore, this critical review summarizes the correct definitions for gas conversion in plasma reactors and highlights common errors and inconsistencies observed throughout literature. This is done for pure CO_2 splitting, dry reforming of methane and CO_2 hydrogenation. We demonstrate that the change in volumetric flow rate is a critical aspect, inherent to these reactions, that is often not correctly taken into account. For dry reforming of methane and CO_2 hydrogenation, we also demonstrate inconsistent reporting of energy efficiency, and through numerical examples, we show the significance of these deviations. Furthermore, we discuss how to measure changes in volumetric flow rate, supported by data from two experimental examples, showing that the sensitivity inherent to a standard component and a flow meter is essential to consider when deriving the performance metrics. Finally, some general recommendations and good practices are provided. This paper aims to be a comprehensive guideline for authors, to encourage more consistent calculations and stimulate the further development of this technology.

Keywords

Plasma; CO₂; Dry reforming of methane; CO₂ hydrogenation; Conversion; Performance

List of symbols

| Symbols (incl. sub- and superscripts) | Description | | | | | | |
|---------------------------------------|---|--|--|--|--|--|--|
| * | Incorrect expression | | | | | | |
| A | Atom | | | | | | |
| abs | Absolute | | | | | | |
| арр | Applied | | | | | | |
| α | Flux ratio | | | | | | |
| b | Balance | | | | | | |
| β | Dilution ratio | | | | | | |
| chem | Chemical | | | | | | |
| conv | Converted | | | | | | |
| corr | Corrected | | | | | | |
| des | Desired | | | | | | |
| DF | Dissociation fraction | | | | | | |
| Δ H ° | Standard reaction enthalpy (kJ mol ⁻¹) | | | | | | |
| E | Energy (kJ ⁻¹ , eV) | | | | | | |
| EC | Energy cost (kJ mol ⁻¹ , kJ L ⁻¹ , eV molecule ⁻¹) | | | | | | |
| eff | Effective | | | | | | |
| EY | Energy yield (mol kJ ⁻¹ , L kJ ⁻¹ , molecule eV ⁻¹) | | | | | | |
| fin | Final | | | | | | |
| | Formation enthalpy (kJ mol ⁻¹) | | | | | | |
| i | Reactant | | | | | | |
| in | At reactor inlet/before reactor | | | | | | |
| init | Initial | | | | | | |
| j | (Gaseous) product | | | | | | |
| k | Condensed/deposited product | | | | | | |
| | Lower heating value (kJ mol ⁻¹) | | | | | | |
| meas | Measured | | | | | | |
| μ | Stoichiometric coefficient | | | | | | |
| <u> </u> | Molar flow (mol min ⁻¹) | | | | | | |
| N _A | Avogadro's constant (molecule mol ⁻¹) | | | | | | |
| η | Energy efficiency | | | | | | |
| out | At reactor outlet/after reactor | | | | | | |
| <i>p</i> | Pressure (bar, atm) | | | | | | |
| P | Power (kW) | | | | | | |
| <i>RE</i> | Relative error | | | | | | |
| reac | Reaction | | | | | | |
| rep | Reported | | | | | | |
| <i>S</i> | Diluting agent/standard component | | | | | | |
| <i>S</i> | Selectivity | | | | | | |
| SEI | Specific energy input (kJ mol ⁻¹ , kJ L ⁻¹ , eV | | | | | | |
| | molecule ⁻¹) | | | | | | |
| <u>I</u> | Tetel | | | | | | |
| tot | Iotal | | | | | | |

| Ϋ́. | Volume flow (L min ⁻¹) |
|----------------|-------------------------------------|
| V _m | Molar volume (L mol ⁻¹) |
| <u>х</u> | Conversion |
| y y | Molar/volume fraction |
| Y | Product yield |

1. Introduction

The transition to a more sustainable society requires innovative management of CO_2 to minimize environmental risks [1]. Carbon capture and utilization (CCU) aims to both capture CO_2 from emission sources, as well as use it as a feedstock for cleaner processes [2]. Various CCU pathways are under development, and a wide variety of methods for chemical CO_2 conversion into, e.g., liquid fuels have been extensively investigated, including thermal, electrochemical or photochemical conversion pathways [3–8].

A CO₂ conversion technology that has gained increasing interest over the past decades is plasma technology [9]. Plasma is an ionized gas that is able to convert stable molecules into value-added chemicals at ambient conditions, e.g., carbon dioxide (CO₂ conversion) [9,10], but also molecular nitrogen (N₂ fixation, for fertilizer production) [11–14]. This technology is very flexible in terms of input gases and instant control over the process, which makes it suitable for electrified production in combination with the fluctuating renewable electricity supply [9]. Furthermore, it does not require using scarce materials [9] and has already been demonstrated in various industrial applications, such as ozone production and arc plasma furnaces for steelmaking [15].

The performance of plasma technology, also with respect to other CCU technologies, has been extensively reported in the review paper by Snoeckx and Bogaerts [9]. As is clear from their work, there is a variety of plasma reactors used for gas conversion, differing in terms of design, applicable currents, voltages, pressures, flow rates, etc. In their paper (as well as in more recent literature, e.g., Refs. [16–19]), a comparison of these reactors in terms of their performance towards CO_2 conversion is presented, which is extremely important for developing this technology. To make this comparison, it is imperative to have an objective, consistent and correct methodology for determining the performance metrics.

Ideally, a plasma reactor can be considered as a plug flow reactor, i.e., a tubular reactor where the composition of the gas is uniform in the radial direction, but changes as a function of the axial position. This also means that the total number of molecules, the concentrations and the total volumetric flow rate change along the reactor. Indeed, a change in volumetric flow rate is inherent to any reaction where the total number of molecules changes and should be considered when determining the performance metrics. For example, in the case of CO_2 splitting, the overall reaction proceeds as follows:

$$\mathrm{CO}_2 \rightleftharpoons \mathrm{CO} + \frac{1}{2}\mathrm{O}_2 \tag{R1}$$

One molecule of reactant splits into 1.5 molecules of product. In other words, upon conversion of CO₂ there will be an increase in the volumetric flow rate. Hence, the performance metrics (such as conversion, product yield and selectivity, but also energy efficiency and cost) cannot be based solely on the number density without considering the change in volumetric flow rate, but should be based on the number of species entering/exiting the reactor per unit of time. These flow rates are well-defined at the inlet of a reactor with mass flow controllers. At the exit of a reactor, however, this is not straightforward to measure. Indeed, common analytical tools (e.g., Fourier transform infrared spectroscopy (FTIR), gas chromatography (GC) and mass spectrometry (MS)) sample the gas flow and measure the concentration of species, but not the volumetric flow rates. To calculate the performance metrics, one should carefully consider whether the flow rate or the concentration is needed.

However, Pinhão et al. [20] found that so-called "gas expansion" (i.e., an increase in the volumetric flow rate) is often neglected when analyzing the performance of plasma reactors. They found errors

up to 20% on the performance metrics for $CO_2/CH_4/He$ mixtures. Therefore, they introduced a new technique to determine the gas expansion ratio, defined as the flux ratio α , alongside a set of formulas to correctly define the performance metrics.

Despite this fundamentally important work, several researchers in the field still neglect gas expansion or provide no information on how it has been taken into account, as will be demonstrated in this paper (see examples and references in Sections 3 and 5). Even when it is accounted for, a wide variety of formulas and interpretations are reported, also within our own group. Furthermore, inconsistent formulas are found in literature for all performance metrics, not only for the conversion, but also for the energy efficiency and energy cost. This complicates the comparison of the results of different works in literature, and hence, further development of this technology.

In this critical review, we aim to clarify the correct definitions for performance metrics in plasma-based gas conversion. In Section 2, we derive the stoichiometric equations for pure CO_2 conversion, and we provide the correct definitions for the energy-related performance metrics. In Section 3, we discuss our equations in relation to other terminology in literature and demonstrate how some other formulas used in literature provide incorrect values, by means of a numerical example. In addition, we illustrate these inconsistencies with a literature review. Section 4 presents the equations for other CO_2 reforming processes, such as dry reforming of methane (DRM) and CO₂ hydrogenation. They typically require more elaborate calculations, since various products can be formed, and simple stoichiometry is typically not sufficient, and thus, the concept of flux ratio should be used. In Section 5, we discuss our equations again in relation to other terminology that exists in literature for various CO₂ reforming processes (with a co-reactant), and show the large inconsistency towards how energy efficiency is defined through a numerical example, as well as through a review of some of the available recent literature. Section 6 presents different analytical methods to obtain the flux ratio, supported by two experimental examples, both for pure CO_2 conversion and with co-reactants, to evaluate their accuracy and choose the best method. Finally, in Section 7, we provide some research recommendations and good practices for plasma-based CO_2 conversion, followed by the overall conclusion in Section 8. It is worth stressing that the presented definitions here are valid for any type of gas conversion in a plug flow reactor and not limited to plasma processes; however, a detailed analysis of the equations used in other gas conversion fields is beyond the scope of this paper.

2. CO₂ conversion: definitions and terminology

Pure CO₂ splitting is the first reaction that we will discuss as a baseline for different reactant mixtures later in Section 4. Thanks to the simple stoichiometry, we can derive all performance metrics directly from the measured output fractions. Besides the theoretical derivation in this section, we will also test these formulas in an experimental example in Section 6.2.

2.1 Conversion

The conversion χ can be deduced from the stoichiometry of the reaction for pure CO₂ splitting, as indicated in Table 1.

Table 1. Reaction equation for pure CO₂ conversion, expressed in flow rates relative to the total inlet flow rate.

| Reaction | CO ₂ | СО | O ₂ |
|----------|-----------------|----|-----------------------|
| in | 1 | 0 | 0 |
| out | $1-\chi$ | χ | χ/2 |

After the reaction, we can express the measured concentration of CO_2 as the CO_2 output fraction $y_{CO_2}^{out}$:

$$y_{\rm CO_2}^{\rm out} = \frac{\dot{n}_{\rm CO_2}^{\rm out}}{\dot{n}_{\rm tot}^{\rm out}} = \frac{\dot{n}_{\rm CO_2}^{\rm out}/\dot{n}_{\rm tot}^{\rm in}}{\dot{n}_{\rm tot}^{\rm out}/\dot{n}_{\rm tot}^{\rm in}} = \frac{1-\chi}{(1-\chi)+\chi+\frac{\chi}{2}} = \frac{1-\chi}{1+\frac{\chi}{2}}$$
(1)

with $\dot{n}_{CO_2}^{out}$ and \dot{n}_{tot}^{out} the CO₂ and total molar flow rate at the reactor outlet, respectively, and \dot{n}_{tot}^{in} the total molar flow rate at the reactor inlet (which in the case of pure CO₂ splitting is equal to the CO₂ molar flow rate at the reactor inlet $\dot{n}_{CO_2}^{in}$). For the other components, we obtain

$$y_{\rm CO}^{\rm out} = \frac{\chi}{1 + \frac{\chi}{2}} \tag{2}$$

$$y_{O_2}^{\text{out}} = \frac{\frac{\chi}{2}}{1 + \frac{\chi}{2}} \tag{3}$$

The conversion is calculated from any of these measured fractions by simply rearranging the equations (see Supporting Information (SI), Section S1). When the output fraction of CO_2 is measured, we calculate the conversion as follows:

$$\chi = \frac{1 - y_{\rm CO_2}^{\rm out}}{1 + \frac{y_{\rm CO_2}^{\rm out}}{2}} \tag{4}$$

This formula inherently accounts for gas expansion and can be calculated similarly from the other measured output fractions, as presented in the SI (Section S1). In the rest of the paper, we name the equations of this section "stoichiometric equations", since they were derived from the stoichiometric reaction.

2.2 Flux ratio

Often, the conversion is presented in terms of the fraction at the inlet (i.e., measured as a "blank" at the outlet when the plasma is OFF) and outlet of the plasma (i.e., measured when the plasma is ON):

$$\chi = \frac{y_{\text{CO}_2}^{\text{in}} - \alpha \cdot y_{\text{CO}_2}^{\text{out}}}{y_{\text{CO}_2}^{\text{in}}}$$
(5)

Here, $y_{CO_2}^{in}$ is the input fraction of CO₂ (which is 1 for pure CO₂) and the factor α is the so-called "flux ratio" introduced by Pinhão et al. [20]. In the case of pure CO₂ splitting, this is also called the "gas expansion factor", and it represents how much the total molar flow rate \dot{n}_{tot}^{out} has increased due to the reaction (i.e., $\dot{n}_{tot}^{out} > \dot{n}_{tot}^{in}$). If the ideal gas law is considered valid, then at the same pressure and temperature, this is equal to the ratio of the total volumetric flow rate at the outlet \dot{V}_{tot}^{out} relative to the inlet \dot{V}_{tot}^{in} :

$$\alpha = \frac{\dot{n}_{\text{tot}}^{\text{out}}}{\dot{n}_{\text{tot}}^{\text{in}}} = \frac{\dot{V}_{\text{tot}}^{\text{out}}}{\dot{V}_{\text{tot}}^{\text{in}}}$$
(6)

Note that it is a general assumption in this work to consider the ideal gas law as valid, as well as that the values for the volumetric flow rates are always obtained at the same temperature and pressure.

For pure CO₂ conversion (with $\dot{n}_{tot}^{in} = \dot{n}_{CO_2}^{in}$), we can couple this back to the stoichiometric equations (Table 1):

$$\alpha = \frac{\dot{n}_{\text{tot}}^{\text{out}}}{\dot{n}_{\text{tot}}^{\text{in}}} = \frac{\dot{n}_{\text{CO}_2}^{\text{out}} + \dot{n}_{\text{CO}_2}^{\text{out}} + \dot{n}_{\text{O}_2}^{\text{out}}}{\dot{n}_{\text{CO}_2}^{\text{in}}} = \frac{(1-\chi) + \chi + \chi/2}{1} = 1 + \frac{\chi}{2}$$
(7)

This result fits the intuitive understanding of gas expansion. Every CO molecule replaces exactly one CO_2 molecule after the reaction, but in addition, half of an O_2 molecule is formed as well. In other words, it is equal to the initial volume of CO_2 (=1) plus half of the converted CO_2 (to account for the "extra" ½ O_2 molecule). If we fill the result of Eq. (7) back into Eq. (5):

$$\chi = \frac{y_{\rm CO_2}^{\rm in} - \left(1 + \frac{\chi}{2}\right) \cdot y_{\rm CO_2}^{\rm out}}{y_{\rm CO_2}^{\rm in}}$$
(8)

$$\chi = \frac{1 - \left(1 + \frac{\chi}{2}\right) \cdot y_{\text{CO}_2}^{\text{out}}}{1} \tag{9}$$

and rearrange to extract the conversion, we obtain the same result as Eq. (4). It is clear that all these Eqs. (1)–(9) are related to one another. Moreover, only one of the parameters must be known to calculate the others if we assume a perfect reaction like pure CO_2 splitting. The relation between all the different parameters is summarized in the SI (Section S1). This approach is similar for all reactions that have no side reactions or no significant amount of by-products, such as NH₃ synthesis from N₂ and H₂ or the formation of H₂O from H₂ and O₂.

2.3 Diluted CO₂

In some cases, a dilution gas, such as argon or helium, is added to the feed gas to improve plasma stability and performance. However, this dilution factor is not always considered when comparing the performance, and thus, the real gas conversion might be overestimated.

To take this dilution of CO₂ into account, we introduce the "dilution ratio" β , expressed as

$$\beta = \frac{\dot{n}_s}{\dot{n}_{\rm CO_2}^{\rm in}} = \frac{\dot{V}_s}{\dot{V}_{\rm CO_2}^{\rm in}} \tag{10}$$

With \dot{n}_s and \dot{V}_s the molar and volumetric flow rate, respectively, of the diluting agent (the subscript *s* refers to "standard component", for which the same derivation of the conversion applies as for the diluting agent, as we will explain in more detail in Section 6.1.3). With this dilution ratio, one can again derive the conversion from any of the measured output fractions, including the fraction of the diluting agent y_s . This is also presented in the SI (Section S2). When the output fraction of CO₂ is measured, we calculate the conversion as follows:

$$\chi^{\text{abs}} = \frac{1 - (1 + \beta) \cdot \gamma_{\text{CO}_2}^{\text{out}}}{1 + \frac{\gamma_{\text{CO}_2}^{\text{out}}}{2}}$$
(11)

The derivation of this equation can also be found in the SI (Section S2). We define this parameter as the absolute conversion, which only considers the conversion of a single reactant of interest. When considering the whole input mixture, we define the effective conversion χ^{eff} :

$$\chi^{\rm eff} = \chi^{\rm abs} \cdot y^{\rm in}_{\rm CO_2} \tag{12}$$

Where the absolute conversion χ^{abs} is multiplied by the fraction of CO₂ at the inlet $y_{CO_2}^{in}$. This definition of the conversion is important for the comparison of different reactant mixtures. It can help in understanding how the conversion of one specific reactant is affected by dilution in different gases. Moreover, the effective conversion is the correct input for calculating the energy cost and energy efficiency, while the absolute conversion would give too optimistic results (underestimated energy cost and overestimated energy efficiency). Indeed, the energy is applied to the full gas mixture, not only to CO₂. We discuss these definitions of interest next.

2.4 Energy cost, energy yield and energy efficiency

The two most important experimental input parameters in any plasma-based gas conversion process are the plasma power and the input flow rate. The plasma power is determined according to the specific plasma type. For example, in a dielectric barrier discharge (DBD) reactor, the area of the Lissajous figures is calculated [21]. In contrast, in a gliding arc reactor, the time-averaged product of voltage and current is taken [22], and in a radiofrequency (RF) or microwave (MW) reactor, the difference between forward and reflected power determines the plasma power [23]. The input flow rate is determined with mass flow controllers (MFCs). It is crucial to pay special attention to the manufacturer and calibration of the MFCs, since the definitions of "standard" and "normal" liter per minute can differ depending on the region or institute; for instance, American standard liters per minute is the same as European normal liters per minute [24–27]. These differences can have a significant effect, as demonstrated with the molar volume in Table 2 when assuming the ideal gas law.

Table 2. Various definitions for standard and normal conditions and the corresponding values for the molar volume V_m . The EPA is the Environmental Protection Agency of the USA. The IUPAC definitions for standard temperature and pressure (STP) and standard ambient temperature and pressure (SATP) are also given.

| Definition | <i>Т</i> (К) | <i>Т</i> (°С) | <i>p</i> (atm) | <i>p</i> (bar) | V _m (L mol⁻¹) |
|--------------------------|--------------|---------------|----------------|----------------|--------------------------|
| normal EU = standard USA | 273 | 0 | 1 | 1.01325 | 22.41 |
| standard EU = normal USA | 293 | 20 | 1 | 1.01325 | 24.06 |
| EPA | 298 | 25 | 1 | 1.01325 | 24.47 |
| STP IUPAC | 273 | 0 | 0.987 | 1 | 22.71 |
| SATP IUPAC | 298 | 25 | 0.987 | 1 | 24.79 |

These two input parameters are combined into the specific energy input *SEI*. The specific energy input is the dominant determining factor for the conversion and energy efficiency in a plasma process and is defined as the ratio of plasma-deposited power *P* to the inlet gas flow rate. *SEI* can be expressed in different units, depending on the units of the power and the inlet gas flow rate, as shown through Eq. (13) in kJ mol⁻¹, (14) in kJ L⁻¹ and (15) in eV molecule⁻¹:

$$SEI (kJ mol^{-1}) = \frac{P (kW)}{\dot{n}_{tot}^{in} (mol min^{-1})} \cdot 60 (s min^{-1})$$
(13)

$$SEI (kJ L^{-1}) = \frac{P (kW)}{\dot{V}_{tot}^{in} (L \min^{-1})} \cdot 60 (s \min^{-1}) = \frac{SEI(kJ \mod^{-1})}{V_m (L \mod^{-1})}$$
(14)

$$SEI (eV molecule^{-1}) = \frac{SEI(kJ mol^{-1}) \cdot 6.24 \cdot 10^{21} (eV kJ^{-1})}{N_A (molecule mol^{-1})}$$
(15)

Where $V_{\rm m}$ is the molar volume of an ideal gas, calculated at the same conditions as the measurement of the flow rate (see Table 2), and N_A is Avogadro's constant (i.e., 6.02214076 x 10²³ mol⁻¹). For clarity, in the following definitions throughout this paper, we assume the specific energy input to be defined in kJ mol⁻¹, as in Eq. (13), but of course the units are readily interchangeable.

Based on the specific energy input, we calculate the energy cost EC as the amount of energy that is consumed by the process:

$$EC (kJ mol_{conv}^{-1}) = \frac{SEI (kJ mol^{-1})}{\chi^{eff}}$$
(16)

As written above, note that the energy cost can also be expressed in kJ L^{-1} and eV molecule⁻¹, depending on the unit of the specific energy input. An alternative is to express the inverse of the energy cost, i.e., the energy yield *EY*:

$$EY \left(\text{mol}_{\text{conv}} \, \text{kJ}^{-1} \right) = \frac{1}{EC \left(\text{kJ} \, \text{mol}_{\text{conv}}^{-1} \right)}$$
(17)

The energy efficiency η is the third main property that describes the effectiveness of the applied energy during the process, next to energy cost and energy yield. It is a measure of how efficiently the process performs compared to the standard reaction enthalpy (+283 kJ mol⁻¹ for pure CO₂ splitting):

$$\eta = \frac{\chi^{\text{eff}} \cdot \Delta H^{\circ} \,(\text{kJ mol}^{-1})}{SEI \,(\text{kJ mol}^{-1})}$$
(18)

3. Confusing terminology and errors in literature for pure CO₂ splitting

In the previous section, we gave an overview of the correct definitions and calculations for determining CO_2 conversion, and the corresponding energy cost and energy efficiency, in a plasma reactor. In this section, we discuss some common definitions used in literature for gas conversion and the errors that they introduce in the reported values (Section 3.1). As an illustration, we present a numerical example and demonstrate the difference between the formulas in Section 3.2. Finally, in Section 3.3, we give an overview of the literature, and we refer to numerous papers that report a wide variety of both correct and incorrect results. Note that in Section 3.1, we only discuss the formulas, while in Section 3.3, we refer to the corresponding papers that use these formulas, for a critical literature analysis.

3.1 Alternative and/or incorrect definitions

3.1.1 Most common definition

First, we discuss the most common definition of conversion in plasma literature, expressed as follows:

$$\chi = \frac{\dot{n}_{\rm CO_2}^{\rm conv}}{\dot{n}_{\rm CO_2}^{\rm in}} = \frac{\dot{n}_{\rm CO_2}^{\rm in} - \dot{n}_{\rm CO_2}^{\rm out}}{\dot{n}_{\rm CO_2}^{\rm in}}$$
(19)

This formula is correct, but the problem is that analytical equipment typically measures concentrations and not molar or volumetric flow rates. As mentioned in Section 0, the latter is easily defined at the inlet, but it is not straightforward to obtain them at the outlet. Often, the following equation is applied instead, using the measured fractions:

$$\chi^* = \frac{y_{\rm CO_2}^{\rm in} - y_{\rm CO_2}^{\rm out}}{y_{\rm CO_2}^{\rm in}}$$
(20)

Although Eq. (19) is correct, it is not correct to simply replace the CO₂ flow rates with the CO₂ fractions, as in Eq. (20), and we indicate the incorrect conversion with an asterisk χ^* . This fallacy can be explained as follows. When we rewrite the concentrations again in terms of molar flow rates, we get an equation that is only valid if \dot{n}_{tot}^{in} and \dot{n}_{tot}^{out} are equal:

$$\chi^* = \frac{\frac{\dot{n}_{\rm CO_2}^{\rm in}}{\dot{n}_{\rm tot}^{\rm in}} - \frac{\dot{n}_{\rm CO_2}^{\rm out}}{\dot{n}_{\rm tot}^{\rm out}}}{\frac{\dot{n}_{\rm CO_2}^{\rm in}}{\dot{n}_{\rm tot}^{\rm in}}}$$
(21)

In the CO₂ splitting reaction, however, the molar flow rates at in- and outlet are not equal, as discussed in Section 0. The total number of moles per unit of time \dot{n}_{tot}^{out} increases with a factor between 1 and 1.5, depending on the amount of CO₂ that is converted. As a result, \dot{n}_{tot}^{in} and \dot{n}_{tot}^{out} are not equal and thus, Eqs. (20) and (21) are false. The problem lies in using the relative values of concentration. Indeed, for every CO₂ molecule that reacts away, it creates an additional dilution of the remaining CO₂ molecules. The correct definition follows from introducing a new factor in Eq. (21) as follows:

$$\chi = \frac{\frac{\dot{n}_{\rm CO_2}^{\rm in}}{\dot{n}_{\rm tot}^{\rm in}} - \left(\frac{\dot{n}_{\rm tot}^{\rm out}}{\dot{n}_{\rm tot}^{\rm in}}\right)\frac{\dot{n}_{\rm CO_2}^{\rm out}}{\dot{n}_{\rm tot}^{\rm out}}}{\frac{\dot{n}_{\rm CO_2}^{\rm in}}{\dot{n}_{\rm tot}^{\rm in}}} = \frac{y_{\rm CO_2}^{\rm in} - \alpha \cdot y_{\rm CO_2}^{\rm out}}{y_{\rm CO_2}^{\rm in}}$$
(22)

where α is the flux ratio, exactly as described in Section 2.2; in other words, Eq. (22) is the same as Eq. (5). If this factor is neglected by simply applying equation (20), the conversion is overestimated, as we will show with a numerical example in Section 3.2.

3.1.2 Measured conversion

A second common definition in literature describes a "corrected" conversion $\chi^{\rm corr}$. The overestimation of equation (20) is thereby often attributed to the sampling procedure of analytical equipment. However, gas expansion is inherent to chemical reactions, as shown by the stoichiometric formulas in Section 2.1. More specifically, Eq. (20) was defined as the "measured" conversion $\chi^{\rm meas}$ that must be recalculated to the corrected conversion. For example, the following equation is common:

$$\chi^{\text{meas}} = \frac{y_{\text{CO}_2}^{\text{in}} - y_{\text{CO}_2}^{\text{out}}}{y_{\text{CO}_2}^{\text{in}}} = 1 - \left(\frac{1 - \chi^{\text{corr}}}{1 + \frac{\chi^{\text{corr}}}{2}}\right)$$
(23)

and can be rearranged to obtain the corrected CO₂ conversion:

$$\chi^{\rm corr} = \frac{2 \cdot \chi^{\rm meas}}{3 - \chi^{\rm meas}} \tag{24}$$

It is important to note that the resulting value is correct, but the explanation is somewhat ambiguous and might be confusing. This is especially true because of the term measured conversion, while we demonstrated in Section 2 that simple stoichiometric formulas suffice to calculate the conversion.

Taking a closer look at Eq. (23), we want to make it clear that this is not a conversion, just like Eq. (20) is not a conversion. Instead, what is calculated is the product fraction:

$$\frac{y_{\rm CO_2}^{\rm in} - y_{\rm CO_2}^{\rm out}}{y_{\rm CO_2}^{\rm in}} = \frac{1 - y_{\rm CO_2}^{\rm out}}{1} = \sum_j y_j^{\rm out}$$
(25)

With *j* representing any product, in this case CO and O₂. Note that Eq. (25) is only valid in case of pure CO₂ because $y_{CO_2}^{in}$ has to be equal to 1. The value of $y_{CO_2}^{out}$ can be derived from the stoichiometry as in Eq. (1):

$$\sum_{j} y_{j}^{\text{out}} = 1 - \frac{1 - \chi}{1 + \frac{\chi}{2}}$$
(26)

Rearranging this equation to extract the conversion χ :

$$\chi = \frac{2 \cdot \sum_{j} y_{j}^{\text{out}}}{3 - \sum_{j} y_{j}^{\text{out}}}$$
(27)

gives the same result as Eq. (24). To verify our new definition, i.e., Eq. (4), we can check that the product fraction, according to the definitions in Section 2, is as follows:

$$\sum_{j} y_j^{\text{out}} = y_{\text{CO}}^{\text{out}} + y_{\text{O}_2}^{\text{out}}$$
⁽²⁸⁾

Inserting Eqs. (2) and (3) in Eq. (28) gives:

$$\sum_{j} y_{j}^{\text{out}} = \frac{\chi}{1 + \frac{\chi}{2}} + \frac{\frac{\chi}{2}}{1 + \frac{\chi}{2}} = \frac{\frac{3 \cdot \chi}{2}}{1 + \frac{\chi}{2}} = \frac{3 \cdot \chi}{2 + \chi}$$
⁽²⁹⁾

After rearranging this equation to extract the conversion, this gives the same expression as equation (27).

3.1.3 Dissociation fraction

The third definition that is commonly applied in literature to express conversion is the dissociation fraction $\chi_{\rm DF}$, as follows:

$$\chi_{\rm DF} = \frac{\gamma_{\rm CO}^{\rm out}}{\gamma_{\rm CO_2}^{\rm out} + \gamma_{\rm CO}^{\rm out}} \tag{30}$$

When these values are replaced by the stoichiometry from Eqs. (1) and (2), this gives

$$\chi_{\rm DF} = \frac{\frac{\chi}{1 + \chi/2}}{\frac{1 - \chi}{1 + \chi/2} + \frac{\chi}{1 + \chi/2}}$$
(31)

The equation can be rearranged as follows:

$$\chi_{\rm DF} = \frac{\chi}{1 - \chi + \chi} = \chi \tag{32}$$

This proves that the dissociation fraction is an alternative to calculate the conversion in pure CO_2 . Of course, this requires an accurate determination of both CO and CO_2 in the analytical equipment, as opposed to the simpler definition of Eq. (4) where only the output CO_2 fraction is needed.

However, a misleading variation of the dissociation fraction is sometimes applied, which we label with an asterisk in the following equations to highlight the difference. The equation goes as follows:

$$\chi_{\rm DF}^* = \frac{y_{\rm CO}^{\rm out}}{y_{\rm CO_2}^{\rm in}} \tag{33}$$

However, if the stoichiometric principles are applied, this gives

$$\chi_{\rm DF}^* = \frac{\frac{\chi}{1+\chi/2}}{1} = \frac{\chi}{1+\chi/2} = \gamma_{\rm CO}^{\rm out}$$
(34)

This is not equal to the conversion; instead, this dissociation fraction expresses the output fraction of CO (Eq. (2)). After rearranging equation (34), the conversion becomes

$$\chi = \frac{\chi_{\rm DF}^*}{1 - \frac{\chi_{\rm DF}^*}{2}} = \frac{\gamma_{\rm CO}^{\rm out}}{1 - \frac{\gamma_{\rm CO}^{\rm out}}{2}}$$
(35)

Hence, as we will show in the next section, authors who apply this formula underestimate their real conversion.

An important observation is that the error made in Eq. (33) is a result of neglecting the change in volumetric flow rate. Indeed, one can derive the following expression from Eq. (19) and Table 1:

$$\chi = \frac{\dot{n}_{\rm CO_2}^{\rm conv}}{\dot{n}_{\rm CO_2}^{\rm in}} = \frac{\dot{n}_{\rm CO}^{\rm out}}{\dot{n}_{\rm CO_2}^{\rm in}} = \frac{\dot{n}_{\rm tot}^{\rm out}}{\dot{n}_{\rm tot}^{\rm out}} \cdot \frac{\dot{n}_{\rm CO}^{\rm out}}{\dot{n}_{\rm CO_2}^{\rm in}} = \alpha \cdot y_{\rm CO}^{\rm out}$$
(36)

This shows that Eq. (33) can only be correct when the flux ratio is 1, which is inherently never the case for CO_2 splitting. Essentially, this is the exact same fundamental error that is made as with Eq. (20), indicating that the aspect of determining the flow rates of the output gases (through the fractions of the output gases and the flux ratio) is still poorly understood by some authors, despite being so important.

3.2 Numerical example

As an illustration, we present a numerical example to compare our correct (stoichiometric) formula from Section 2 (Eq. (4)) with the alternative formulas from Section 3.1, and demonstrate which formulas give the same or different results as our stoichiometric calculation. We consider CO_2 conversions in a wide range from 10% to 90%. From this range, we calculate the conversion according to the various definitions and present these in Table 3. We also plot the reported conversions as a function of the real conversion in Fig. 1, where it is clear that the stoichiometric conversion from Eq. (4) forms a perfectly straight line, as it is the correct method for calculating the conversion. This is also true for Eqs. (27) and (30), as explained in Section 3.1. However, Eqs. (20) and (33) deviate. The dashed lines indicate the relative error of these other definitions, as denoted by the *y*-axis on the right. Indeed, Eq. (20) overestimates the conversion, while Eq. (33) underestimates it. Note that the relative error on the conversion RE_{χ} , similar to the work of Pinhão et al. [20], is calculated as follows:

$$RE_{\chi} = \frac{\chi^{\text{rep}} - \chi}{\chi} \times 100 \ (\%) \tag{37}$$

Where χ^{rep} is the reported conversion and χ is the real conversion. If the reported conversion overestimates the real conversion, the relative error is positive; on the contrary, if it is an underestimation, the relative error is negative. It is clear from Fig. 1 that the relative errors of Eq. (20) are larger for smaller conversions, i.e., a relative error of 40% is obtained for a conversion of 10%. In Eq. (33), the relative error is the largest (most negative) at higher conversions, i.e., a relative error of - 25% is reached for a conversion of 70%. Furthermore, these errors propagate to other performance metrics, such as the energy efficiency and energy cost.

This example demonstrates the importance of using the correct formulas for consistency when comparing the performance of plasma-based CO_2 conversion in different types of plasma reactors, and more in general, for moving this technology forward.



Fig. 1. Numerical example of the reported conversion calculated according to the different equations (see legend; full lines, left axis) and the relative error (dotted lines, right axis). Since the correct formulas report the real CO_2 conversion, their relative error is equal to zero. Eq. (20) yields a positive error, while Eq. (33) gives a negative relative error.

| | | Measured | fractions | | Stoichi conver | ometric sion (4) | (2 | 0) | (2 | 7) | (3 | 0) | (3 | 3) |
|------------------------------------|-------------------------|--------------------------|------------------------|-----------------|-------------------|---|-------------------------------|------------------------------------|-------------------------------------|--------------------|--|-----------------------|--|---------------------|
| Real CO ₂ conversion | $y_{\rm CO_2}^{\rm in}$ | $y_{\rm CO_2}^{\rm out}$ | $y_{\rm CO}^{\rm out}$ | $y_{0_2}^{out}$ | $\frac{1-}{1+}$ | $\frac{y_{\rm CO_2}^{\rm out}}{y_{\rm CO_2}^{\rm out}}$ | $\frac{y_{CO_2}^{in}}{y_C^i}$ | $- \frac{y_{CO_2}^{out}}{n}_{O_2}$ | $\frac{2 \cdot \Sigma}{3 - \Sigma}$ | y_j^{out} | $\frac{y_{\rm C}^{\rm o}}{y_{\rm CO_2}^{\rm out}}$ | x_{CO}^{out} | $\frac{y_{\rm C}^{\rm c}}{y_{\rm C}^{\rm in}}$ | n O ₂ |
| 10.0 | 100 | 85.7 | 9.5 | 4.8 | 10.0 | = | 14.3 | + | 10.0 | = | 10.0 | = | 9.5 | |
| 20.0 | 100 | 72.7 | 18.2 | 9.1 | 20.0 | = | 27.3 | + | 20.0 | = | 20.0 | = | 18.2 | |
| 30.0 | 100 | 60.9 | 26.1 | 13.0 | 30.0 | = | 39.1 | + | 30.0 | = | 30.0 | = | 26.1 | l |
| 40.0 | 100 | 50.0 | 33.3 | 16.7 | 40.0 | = | 50.0 | + | 40.0 | = | 40.0 | = | 33.3 | |
| 50.0 | 100 | 40.0 | 40.0 | 20.0 | 50.0 | = | 60.0 | + | 50.0 | = | 50.0 | = | 40.0 | - |
| 60.0 | 100 | 30.8 | 46.2 | 23.1 | 60.0 | = | 69.2 | + | 60.0 | = | 60.0 | = | 46.2 | I |
| 70.0 | 100 | 22.2 | 51.9 | 26.9 | 70.0 | = | 77.8 | + | 70.0 | = | 70.0 | = | 51.6 | l |
| 80.0 | 100 | 14.3 | 57.1 | 28.6 | 80.0 | = | 85.7 | + | 80.0 | = | 80.0 | = | 57.1 | |
| 90.0 | 100 | 6.9 | 62.1 | 31.0 | 90.0 | = | 93.1 | + | 90.0 | = | 90.0 | = | 62.1 | _ |

Table 3. Numerical example of the conversion as calculated by the different equations. The first row gives the reference to the equation number in the paper, while the second row gives the symbols and formulas. The comparison to the real conversion is indicated as overestimated (+), equal (=) or underestimated (-) values. For clarity, all values are given in %.

3.3 Critical evaluation of literature reports

While the example in Section 3.2 highlights the importance of applying the correct formulas, we illustrate the lack of uniformity in this section with some examples from literature.

Luckily, many recent studies on plasma-based CO₂ conversion do report the correct values for conversion and energy efficiency, although the simple stoichiometric derivation (Eq. (4)) is rarely applied (e.g., in Refs. [28,29]). Many authors, including from our own group, e.g., Refs. [22, 30-35], apply the confusing terminology of "measured" conversion and then "corrected" this conversion in the next step (i.e., Eq. (27) of Section 3.1). In the present work, we demonstrated in Eqs. (25)–(29) that it is not a "measured" conversion, but the product fraction $\sum_j y_j^{\text{out}}$ that can be used to calculate the real CO₂ conversion χ . Although the definitions are confusing, the reported values remain correct.

Another common definition is the dissociation fraction (as in Eq. (30)), which is indeed equivalent to the conversion, see Refs. [36–38]. Various other derivations for the CO_2 conversion can be found in literature (e.g., Refs. [39–46]). Despite these inconsistent explanations, the formulas can be validated when applying the simple stoichiometric principles from Section 2.1. Hence, the reported values are correct, but for transparency and consistency reasons, we strongly recommend using the simple stoichiometric Eq. (4), or one of the other stoichiometrically derived expressions (SI; Section S1) in future work.

Unfortunately, some papers in the field report overestimated values for the CO₂ conversion when they apply the incorrect equation (20) (e.g., Refs. [47–50]). As demonstrated in Section 3.1, these studies report the product fraction, which is not equal to the conversion (Eq. (27)). Some other sources underestimate their CO₂ conversion when they apply an alternative form of the dissociation fraction (Eq. (33)) (e.g., Refs. [51,52]). In addition, some authors claimed that the GC sampling method was responsible for the deviation in the "measured" conversion. This incorrect explanation in previous work from our group [30] has led to misunderstandings in later work. In one example, also from our group [47], it is even mistakenly declared that gas expansion does not play a role in their analytical method, even though it is inherent to the reaction and independent of the setup. Another observation is that, very often, the correct formula is given (based on molar flow rates), but the reported diagnostics, such as GC, are not able to measure the molar flow rates directly, and no details are given on how these values were obtained (e.g., Refs. [53–59]).

Several authors derive the flux ratio through a flow meter [60–64] or a standard component (either as part of the feed gas or introduced in the exhaust gas stream) [65–69], and do not use a stoichiometrically derived expression, but, e.g., Eq. (5) or (19). Some authors also use a combination of both a stoichiometrically derived expression and a flow meter, to validate their approach [42,70]. These methods will be discussed and compared with each other in more detail in Section 6.

Finally, accounting for the dilutions is important. Many studies (e.g., Refs. [61, 71-74]) report the absolute conversion, and it has been demonstrated that, e.g., the absolute CO_2 conversion increases in the presence of argon [75]. While this is not an error, we do advise to report the effective conversion as well, as this value takes into account the dilution of CO_2 and hence, is the value that should be compared with other reactors.

This section has clearly demonstrated how the correct formulas are essential to analyze the performance of different plasma reactors. Although inconsistent formulas are not always a problem when authors are transparent in their data and calculations, and do not always change the conclusions significantly, they become problematic when authors compare their results to other experiments, as we demonstrated with significant deviations in a numerical example. Therefore, we hope that this

paper can elucidate the correct formulas for CO_2 conversion and help authors to report more consistent, correct values for the performance of plasma reactors.

4. CO₂ conversion with a co-reactant: definitions and terminology

In the previous section, we described the conversion of pure CO_2 , but many recent studies add a coreactant to the feed to facilitate the conversion. Indeed, the reaction enthalpy of pure CO_2 splitting $(\Delta H^\circ = +283 \text{ kJ mol}^{-1})$ is higher than the enthalpy of DRM ($\Delta H^\circ = +247 \text{ kJ mol}^{-1}$), for example. Moreover, the addition of co-reactants allows for direct targeting of valuable reaction products like syngas or oxygenates, decreasing the number of subsequent processing steps like the Fischer-Tropsch synthesis, and therefore lowering the target for the energy efficiency [9].

The presence of the co-reactant does not change the definition of α as written in Eq. (6), but it will certainly affect its value. For example, the DRM reaction is usually expressed with reaction (R2):

$$CO_2 + CH_4 \rightleftharpoons 2 CO + 2 H_2$$
 (R2)

As can be seen, the gas flow will expand upon increasing conversion, with a theoretical maximum flux ratio of 2, i.e., higher than the maximum value of 1.5 in the case of pure CO_2 splitting.

Another example is plasma-based CO_2 hydrogenation. The most common reactions are the CO_2 methanation reaction (R3) and the reverse water gas shift (RWGS) reaction (R4):

$$CO_2 + 4 H_2 \rightleftharpoons CH_4 + 2 H_2 O$$
 (R3)

$$CO_2 + H_2 \rightleftharpoons CO + H_2O \tag{R4}$$

In this case, the total volumetric flow rate will not increase but decrease, or at least remain the same. Hence, the value for the flux ratio will be lower than or equal to 1.

The change in volumetric flow rate is important for these reaction mixtures, since the gas can contract or expand even more than for pure CO_2 . As already pointed out by Pinhão et al. [20] for $CO_2/CH_4/He$ mixtures, neglecting the effect of gas expansion or contraction can cause significant errors. As indicated in their paper, several authors reported incorrect values, and some authors still do not report all performance metrics correctly, as illustrated in Section 5.3.

In the following section, we summarize how to derive these performance metrics, specifically for the most studied reactions in plasma, i.e., DRM and CO₂ hydrogenation. Nevertheless, almost all formulas are generally applicable. The chemistry inside the plasma might be complex, but the formulas to calculate the overall performance metrics, such as conversion, product selectivity and yield, are valid for all reactions between two or more gases in a plug flow reactor (as described in Section 0).

It should be noted that in some cases, a reactant that is not in the gas phase is used, which also affects the experimental methods. An example is the reverse-Boudouard reaction, which describes the conversion of CO_2 together with solid carbon, and we present the correct formulas for this case in the SI (see Section S3). Again, the latter formulas generally apply for every reaction between one or more gases and a solid component in a plug flow reactor.

4.1 Change in total gas flow rate

Although Eq. (6) for the flux ratio remains valid, deriving a specific formula for mixtures of CO_2 with a co-reactant is more complex than for pure CO_2 splitting (R1). Some critical considerations influence the value of the flux ratio. Firstly, reactions (R2), (R3) and (R4) only represent the ideal reaction, but many researchers report several by-products, depending on the exact plasma conditions.

Moreover, the stoichiometric derivation of the flux ratio α is not always possible, but only in some specific cases (e.g., Section 6.3). This is due to liquid or solid products formed during the reaction. Water, methanol, etc. could condense at the reactor walls or gas lines, and sometimes a cold trap deliberately initiates the condensation of liquids. In addition, carbon deposition is typical in CO₂/CH₄ plasmas with a high CH₄ content. For both liquid and solid products, it is often impossible to trace back their original fraction in the gas stream, since they accumulate over time. Even when they remain in the gas phase, components like water are usually difficult to quantify with common analytics like GC. Because of these effects, deriving a clear stoichiometric relationship can be challenging, and in many cases impossible. This is in contrast to pure CO₂ splitting, where it is clear from Section 2 that stoichiometric equations can always be derived.

Therefore, other methods are applied to determine the flux ratio, such as using a volumetric flow meter, or introducing a standard component in the outlet gas stream. These experimental methods will be discussed in more detail in Section 6.1. To first give a more streamlined overview of the correct formulas, we will treat the flux ratio as a known parameter in the following sections.

4.2 Conversion

In both DRM and CO₂ hydrogenation, an additional reactant next to CO₂ is present, so Eq. (5) changes to the more general equation form Eq. (11). Depending on the reaction, the reactant *i* can be CO₂, CH₄ (in case of DRM), or H₂ (in case of CO₂ hydrogenation). The effective conversion as defined in Eq. (12) remains the same. Additionally, the total conversion χ^{tot} is introduced as the sum of all the effective conversions:

$$\chi^{\text{tot}} = \sum_{i} \chi_{i}^{\text{eff}} \tag{38}$$

The advantage of Eq. (38) is that the overall conversion of the entire gas mixture can be quantified and compared. Note that in Section 2.3, the effective and total conversion is the same, since the dilution is a consequence of an inert gas component that is (virtually) not converted, as opposed to these reaction mixtures where multiple components are converted.

4.3 Energy cost, energy yield and energy efficiency

The definitions from Section 2.4 for the specific energy input, energy cost and energy yield remain valid. The only difference here in terms of energy cost and energy yield is that the total conversion should be used, because it represents the overall conversion of the gas mixture:

$$EC (kJ mol_{conv}^{-1}) = \frac{1}{EY(mol_{conv} kJ^{-1})} = \frac{SEI (kJ mol^{-1})}{\chi^{tot}}$$
(39)

The correct formula to calculate the energy efficiency is a bit more complicated. In theory, Eq. (18) can be used again, with the corresponding reaction enthalpy for DRM or CO_2 hydrogenation. However, as we will discuss in detail in Section 5, this equation is not always ideal to use for DRM and CO_2 hydrogenation. Hence, an alternative definition is introduced, expressing the energy efficiency η (also sometimes called the "energy conversion efficiency") as the fraction of the total energy that goes to the formation and breaking of chemical bonds. In other words, it is the fraction of energy that effectively carries out the reaction and is not lost (e.g., as heat). In general, this is written as

$$\eta = \frac{\sum_{j} E_{j}^{\text{chem}}}{E^{\text{app}} + \sum_{i} E_{i}^{\text{chem}}}$$
(40)

Here, E^{app} is equal to the applied energy to the system, which can be determined through the measured applied power. $\sum_i E_i^{chem}$ and $\sum_j E_j^{chem}$ are the chemical energy, intrinsically present in the reactants *i* and the products *j*, respectively. However, while a reaction enthalpy can be measured, e.g., with a calorimeter, the energy intrinsically present in the reactants and products cannot. To solve this issue, another definition represents the chemical energy of products and reactants by a lower heating value *LHV*, which is a measure for the energy that is released upon full oxidation of the specified species. This is the so-called "fuel energy efficiency":

$$\eta = \frac{\alpha \cdot \sum_{j} \left(y_{j}^{\text{out}} \cdot LHV_{j}(\text{kJ mol}^{-1}) \right)}{SEI (\text{kJ mol}^{-1}) + \sum_{i} \left(y_{i}^{\text{in}} \cdot LHV_{i}(\text{kJ mol}^{-1}) \right)}$$
(41)

with LHV_i and LHV_j the lower heating value of reactant *i* and product *j*, respectively. As mentioned, this value represents the reaction enthalpy for the conversion of *i* or *j* upon reacting with O₂ to their most thermodynamically stable products [20,76]. In this case, these products are CO₂ and/or H₂O, which by definition have a lower heating value equal to 0 kJ mol⁻¹, together with O₂. Note that Eq. (41) still resembles the general form of Eq. (40), with the specific energy input representing the applied energy and the lower heating values the chemical energies, all in units of kJ mol⁻¹. The fact that the reaction enthalpy upon full oxidation is used for all reactants and products and, consequently, the energy efficiency of the reaction. Finally, note that lower heating values describe the reaction enthalpy with the produced H₂O considered to remain in the gas phase. When considering H₂O in the liquid phase, and thus taking into account the additional energy released upon condensation, higher heating values for the different components are used instead.

When all products are included (condensed and deposited products as well), Eq. (42) should be used

$$\eta = \frac{\alpha^{\text{fin}} \cdot \sum_{j} \left(y_{j}^{\text{out}} \cdot LHV_{j}(\text{kJ mol}^{-1}) \right) + \alpha^{\text{init}} \cdot \sum_{k} \left(y_{k}^{\text{out}} \cdot LHV_{k}(\text{kJ mol}^{-1}) \right)}{SEI (\text{kJ mol}^{-1}) + \sum_{i} \left(y_{i}^{\text{in}} \cdot LHV_{i}(\text{kJ mol}^{-1}) \right)}$$
(42)

Where the condensed and deposited products are indicated by the subscript k. Note that we now have two flux ratios. When we have condensation or deposition of products from the gas stream, the total number of molecules in the gas flow will decrease. This means a decrease of the final gas flow rate from $\dot{V}_{tot}^{out,init}$ to $\dot{V}_{tot}^{out,fin}$, and thus, following Eq. (6), a decrease in flux ratio from α^{init} to α^{fin} . The relationship between α^{init} (with all products still in the gas phase) and α^{fin} (after condensation and/or deposition of certain products) is given by Eq. (43)

$$\alpha^{\text{init}} = \frac{\alpha^{\text{fin}}}{(1 - \sum_k y_k^{\text{out}})} \tag{43}$$

When the only product that is significantly lost is H_2O ($LHV_{H_2O} = 0 \text{ kJ mol}^{-1}$), it does not require quantification in order to solve Eq. (42). However, if other components such as methanol, ethanol etc. are a significant part of the condensed fraction, they need to be quantified. Specifically, the fraction of these components while they were still in the gas stream needs to be traced back, which is practically impossible. However, alternative equations are available in literature where this is not strictly needed, which we will discuss in more detail in Section 5.1.

4.4 Product selectivity and yield

Aside from conversion, product yield and selectivity are two other critical performance metrics. As discussed before, the conversion states how much of a reactant has reacted away. Product selectivity expresses how much of one specific product is formed, relative to all other products. The product yield is the combination of conversion and product selectivity, and shows how much a particular product is formed, relative to the theoretical maximum amount that could have been formed. When there are no significant side reactions, as in pure CO_2 splitting, expressing a selectivity or yield is not necessary. For (R1), all CO_2 is converted into CO and O_2 (with negligible amounts of other products, such as O_3), and thus, there is a fixed selectivity towards CO and O_2 . The yield of CO or O_2 is not fixed, but it gives no extra information compared to the conversion.

On the other hand, by-products can be formed in case of DRM and CO_2 hydrogenation. Therefore, it is more interesting to report values that express the degree to which a desired product is produced, and even necessary when comparing conditions in one setup or comparing setups with each other. The atom-based (*A*-based, in this case either the carbon-, hydrogen- or oxygen-based) selectivity for product *j* is defined as

$$S_j^A = \frac{\mu_j^A \cdot \alpha \cdot y_j^{\text{out}}}{\sum_i \left(\mu_i^A \cdot \left(y_i^{\text{in}} - \alpha \cdot y_i^{\text{out}}\right)\right)}$$
(44)

With μ_i^A and μ_j^A the stoichiometric coefficients (i.e., the number of atoms A per reactant i and product j, respectively). The sum of the selectivities for the same base-atoms should be 100% when all products are considered, because it represents the distribution of the atoms among the products that are formed.

The atom-based yield of product *j* is defined as

$$Y_j^A = \frac{\mu_j^A \cdot \alpha \cdot y_j^{\text{out}}}{\sum_i (\mu_i^A \cdot y_i^{\text{in}})}$$
(45)

In this case, the sum of the yields for the same base-atom should be equal to the conversion of the reactants containing the same base-atom, when all products are considered. Indeed, product yield expresses the actual amount of a product relative to the theoretical maximum amount that can be formed of that same product. Mathematically, it can be considered as the product of conversion and selectivity for the same base-atom:

$$Y_{j}^{A} = S_{j}^{A} \cdot \chi_{i}^{A} = \frac{\mu_{j}^{A} \cdot \alpha \cdot y_{j}^{\text{out}}}{\sum_{i} \left(\mu_{i}^{A} \cdot \left(y_{i}^{\text{in}} - \alpha \cdot y_{i}^{\text{out}} \right) \right)} \cdot \frac{\sum_{i} \left(\mu_{i}^{A} \cdot \left(y_{i}^{\text{in}} - \alpha \cdot y_{i}^{\text{out}} \right) \right)}{\sum_{i} \left(\mu_{i}^{A} \cdot y_{i}^{\text{in}} \right)}$$
(46)

When there are condensed products, selectivity and yield can be expressed in the same way, but then a distinction between the initial and final flux ratio must be made, just as was the case for Eq. (42). However, when the lost products cannot be quantified (or even qualitatively detected), the selectivities and yields towards all the other products only require knowledge of the final flux ratio (which can be experimentally measured) and they can therefore still be obtained.

5. Confusing terminology and errors in literature for CO₂ conversion with a co-reactant

In the previous section, we presented the correct definitions and calculations for determining the performance metrics in plasma-based CO₂ conversion with a co-reactant. In this section, we discuss some alternative and some incorrect definitions reported in literature for energy cost, energy efficiency and product yield (Section 5.1). As an illustration, we present a numerical example in Section 5.2 and demonstrate the difficult interpretation and comparison of the reported values. Finally, in Section 5.3, we provide a critical overview of some of the available literature.

5.1 Alternative and/or incorrect definitions

5.1.1 Energy cost

The energy cost in DRM is sometimes defined as a "syngas energy cost":

$$EC_{\rm syngas}(\rm kJ\ mol^{-1}) = \frac{SEI(\rm kJ\ mol^{-1})}{\alpha \cdot \left(y_{\rm CO}^{\rm out} + y_{\rm H_2}^{\rm out}\right)}$$
(47)

This value should be interpreted as the amount of energy needed to form a certain amount of syngas. Note that this definition, where the amount of the desired product is used as reference, is quite common for other reactions too, like NH_3 or NO_x formation [23, 77, 78], and is also interesting to report from an economic perspective. However, it is important to notice that Eq. (47) will never give the same result as Eq. (39), even for "ideal" DRM as described by reaction (R2), due to the increase in flow rate:

$$EC_{\rm syngas}(\rm kJ\ mol^{-1}) = \frac{SEI(\rm kJ\ mol^{-1})}{\alpha \cdot \left(y_{\rm CO}^{\rm out} + y_{\rm H_2}^{\rm out}\right)} = \frac{SEI(\rm kJ\ mol^{-1})}{2 \cdot \chi^{\rm tot}} = \frac{EC(\rm kJ\ mol^{-1})}{2}$$
(48)

with Eq. (48) only valid for ideal DRM, where only syngas is formed and the syngas flow rate equals the total inlet flow rate multiplied with a factor $(2 \cdot \chi^{tot})$.

5.1.2 Energy efficiency

In many cases (see references in Section 5.3), an adapted version of Eq. (41) expresses the energy efficiency. The first variant is Eq. (49):

$$\eta = \frac{\alpha \cdot \sum_{j} \left(y_{j}^{\text{out}} \cdot LHV_{j}(\text{kJ mol}^{-1}) \right)}{SEI (\text{kJ mol}^{-1}) + \sum_{i} \left(\left(y_{i}^{\text{in}} - \alpha \cdot y_{i}^{\text{out}} \right) \cdot LHV_{i}(\text{kJ mol}^{-1}) \right)}$$
(49)

The amount of converted reactants is now written in the denominator, instead of the initial fraction. This does not fully correspond with the general definition (see Eq. (40)). Instead, Eq. (49) defines the fraction of the "transformed" energy that carries out the reaction, while the remaining fraction of "transformed" energy is equal to the energy lost as heat. In other words, this equation gives a (theoretical) value of 1 when the reaction proceeds without heat losses, independent of how much reactant is converted.

A second (and third) variant is when only the desired products (represented by subscript *j*, des) are considered (i.e., syngas for DRM, CH_4 or CO for CO_2 hydrogenation), similar to the syngas energy cost in Section 5.1.1. This can be expressed both relative to the total energy at the inlet (as in Eq. (41)) as well as the "transformed" energy (as in Eq. (49)). Both variants are defined through Eqs. (50) and (51), respectively:

$$\eta = \frac{\alpha \cdot \sum_{j,\text{des}} \left(y_{j,\text{des}}^{\text{out}} \cdot LHV_{j,\text{des}}(\text{kJ mol}^{-1}) \right)}{SEI (\text{kJ mol}^{-1}) + \sum_{i} \left(y_{i}^{\text{in}} \cdot LHV_{i}(\text{kJ mol}^{-1}) \right)}$$
(50)

$$\eta = \frac{\alpha \cdot \sum_{j, \text{des}} \left(y_{j, \text{des}}^{\text{out}} \cdot LHV_{j, \text{des}}(\text{kJ mol}^{-1}) \right)}{SEI (\text{kJ mol}^{-1}) + \sum_{i} \left(\left(y_{i}^{\text{in}} - \alpha \cdot y_{i}^{\text{out}} \right) \cdot LHV_{i}(\text{kJ mol}^{-1}) \right)}$$
(51)

Note that Eqs. (50) and (51) consider the chemical energy attributed to the by-products also as a form of "lost" energy. This offers the advantage that, when not all by-products can be quantified, these equations can still be solved.

Indeed, Eqs. (41), (49), (50) and (51) are all correct, but should be interpreted differently nonetheless. Simply stated, an energy efficiency of 1 (or 100%) would mean for:

- Eq. (49) : a reaction that proceeds without heat losses
- Eq. (41) : a reaction that proceeds without heat losses and 100% conversion
- Eq. (51) : a reaction that proceeds without heat losses and 100% product selectivity towards the desired products
- Eq. (50) : a reaction that proceeds without heat losses and 100% product yield of the desired products

Another alternative definition for the energy efficiency is based on the standard formation enthalpies, also sometimes called the "chemical energy efficiency", for each component:

$$\eta = \frac{\alpha \cdot \sum_{j} \left(y_{j}^{\text{out}} \cdot H_{\text{f},j}(\text{kJ mol}^{-1}) \right) - \sum_{i} \left(\chi_{i}^{\text{eff}} \cdot H_{\text{f},i}(\text{kJ mol}^{-1}) \right)}{SEI \text{ (kJ mol}^{-1})}$$
(52)

 $H_{f,i}$ and $H_{f,j}$ are the standard formation enthalpies of the various reactants *i* and products *j*, expressed in kJ mol⁻¹. In fact, Eq. (52) does not correspond to the general definition presented through Eq. (40). Rather, Eq. (52) compares the energy difference corresponding to the reaction, relative to the applied energy:

$$\eta = \frac{\Delta E^{\text{reac}}}{E^{\text{app}}} \tag{53}$$

Note that Eq. (18) also corresponds to this general equation. For an endothermic reaction, this ratio is a correct representation of the energy efficiency (i.e., 0 when no reaction is proceeding, and 1 when all applied energy is absorbed and used to carry out the reaction). However, as is clear from comparing Eq. (40) with Eq. (53), the chemical energy efficiency does not represent a "fraction" of the total energy that carries out the reaction. Additionally, the chemical energy efficiency on the one hand (Eq. (52)) and the fuel energy efficiency on the other hand (Eqs. (41), (49), (50) and (51)) describe the "chemical energy" differently. In fact, for an exothermic reaction (such as reactions (R3) and (R4) when considering H_2O in its liquid state) the chemical energy efficiency will become negative, even when a relatively good conversion and energy cost is obtained. This makes the results from Eq. (52) sometimes difficult to interpret without any additional information on other performance metrics.

In Eq. (52), the reaction enthalpy ΔH° is constructed from the individual formation enthalpies, following Hess's law. However, a consequence of using a reaction enthalpy is that one has to include all products and reactants to determine it correctly, including all the liquid products that are present

in non-negligible amounts. This makes the chemical energy efficiency not always applicable, in contrast to (some of) the fuel energy efficiency equations. In addition, often a single reaction enthalpy is taken from literature and used, similar to Eq. (18), instead of deriving it through the formation enthalpies. For example, for DRM often the reaction enthalpy of 247.3 kJ mol⁻¹ is used. However, this value is only correct for 1 mole CO_2 reacting with 1 mole CH_4 , forming only syngas, as shown in reaction (R2). Any deviation from this will alter the value for the reaction enthalpy. Hence, this approach is incorrect in all cases, except one theoretically ideal situation.

Finally, note that for pure CO₂ splitting, all equations will lead to the same value (given by Eq. (18)):

$$\eta = \frac{\alpha \cdot \left(y_{CO}^{out} \cdot LHV_{CO}(kJ \text{ mol}^{-1}) + y_{O_2}^{out} \cdot LHV_{O_2}(kJ \text{ mol}^{-1})\right)}{SEI(kJ \text{ mol}^{-1}) + y_{CO_2}^{in} \cdot LHV_{CO_2}(kJ \text{ mol}^{-1})}$$
(54)
$$= \frac{\chi_{CO_2}^{eff} \cdot 283 \text{ kJ mol}^{-1}}{SEI(kJ \text{ mol}^{-1})}$$

$$\eta = \frac{\alpha \cdot \left(y_{CO}^{out} \cdot LHV_{CO}(kJ \text{ mol}^{-1}) + y_{O_2}^{out} \cdot LHV_{O_2}(kJ \text{ mol}^{-1}) \right)}{SEI(kJ \text{ mol}^{-1}) + \left(y_{CO_2}^{in} - \alpha \cdot y_{CO_2}^{out} \right) \cdot LHV_{CO_2}(kJ \text{ mol}^{-1})}$$
(55)
$$= \frac{\chi_{CO_2}^{eff} \cdot 283 \text{ kJ mol}^{-1}}{SEI(kJ \text{ mol}^{-1})}$$

$$\eta = \frac{\alpha \cdot y_{CO}^{out} \cdot H_{f,CO}(kJ \text{ mol}^{-1}) + \alpha \cdot y_{O_2}^{out} \cdot H_{f,O_2}(kJ \text{ mol}^{-1}) - \chi_{CO_2}^{eff} \cdot H_{f,CO_2}(kJ \text{ mol}^{-1})}{SEI(kJ \text{ mol}^{-1})} = \frac{\chi_{CO_2}^{eff} \cdot 283 \text{ kJ mol}^{-1}}{SEI(kJ \text{ mol}^{-1})}$$
(56)

Hence, the correct determination of the energy efficiency depends mostly on the correct determination of the effective CO_2 conversion, which we described in detail in Sections 2.3 and 2.4.

5.1.3 Product yield

For product yield there is also an alternative, but incorrect definition:

$$Y_j^{A^*} = \sum_i \left(\mu_i^A \cdot \chi_i^{\text{eff}}\right) \cdot \frac{1}{\mu_j^A} \cdot S_j^A$$
⁽⁵⁷⁾

This equation uses the product of conversion and selectivity to calculate the yield, as was described in Section 4.4. However, when implementing Eqs. (12) and (44) in Eq. (57), we get

$$Y_j^{A^*} = \alpha \cdot y_j^{\text{out}} = \frac{\dot{n}_{\text{tot}}^{\text{out}}}{\dot{n}_{\text{tot}}^{\text{in}}} \cdot \frac{\dot{n}_j^{\text{out}}}{\dot{n}_{\text{tot}}^{\text{out}}} = \frac{\dot{n}_j^{\text{out}}}{\dot{n}_{\text{tot}}^{\text{in}}}$$
(58)

This is not in line with the general definition of yield, which considers the theoretical maximum amount of product *j* that could have been formed (Eq. (45)). Yield depends on the amount of atoms *A* both per molecule of product *j* and per molecule of reactant *i*, while Eq. (58) is independent of this, and therefore incorrect (e.g., for DRM, \dot{n}_{tot}^{in} is not by definition equal to the theoretical maximum amount that can be formed of H₂, as CO₂ will not contribute to its formation).

5.2 Numerical example

As discussed in Section 5.1, the determination of energy efficiency in literature is ambiguous. Different definitions can be theoretically correct, but result in very different values nonetheless. In this section, we present a numerical example to illustrate the variety in results. Specifically, we consider a DRM experiment, based on previous research from our group [79], but the values are generally valid for any warm plasma (e.g., gliding arc, microwave, etc.); see Table 4. We calculate the energy efficiency according to the various definitions, as presented in Fig. 2. An additional example for CO₂ hydrogenation, with values typical for a cold plasma, is presented in the SI, along with more details regarding the calculations for both examples (see SI, Section S4).

Table 4. Numerical example for DRM in a warm plasma with a specific energy input of 240.6 kJ mol⁻¹; flow rates of gases at the inlet, and unreacted gases and products at the outlet.

| Flow rate (mL min ⁻¹) | CO2 | CH₄ | H₂ | со | C_2H_2 | C_2H_4 | C_2H_6 | H₂O | Total |
|--------------------------------------|-----|-----|-----|-----|----------|----------|----------|-----|-------|
| In | 650 | 350 | 0 | 0 | 0 | 0 | 0 | 0 | 1000 |
| Out | 300 | 90 | 375 | 575 | 15 | 2 | 0.4 | 125 | 1482 |



Fig. 2. Numerical example of the energy efficiency calculated according to the various definitions for DRM. Eq. (52)* represents the chemical energy efficiency without taking all products into account, Eq. (52)** represents the chemical energy efficiency when the reaction enthalpy corresponding to the "ideal" stoichiometry is used (described by reaction (R2) for DRM).

As can be seen, the results vary significantly depending on the applied definition, for both the DRM example (Fig. 2) and the CO_2 hydrogenation example (see SI, Section S4). Note again that all fuel energy efficiency formulas (Eqs. (41), (49), (50) and (51)) are correct, if interpreted correctly. Indeed, referring back to Section 5.1.2, some definitions depend on e.g., the conversion, while some do not. As a result, the definition yielding the highest value is Eq. (49), which expresses the fraction of energy that is not lost as heat, and the definition giving the lowest value is Eq. (50), which expresses the fraction of energy that is not lost as heat, unconverted reactants and by-products.

A large drop in energy efficiency is observed when comparing the fuel energy efficiencies to the chemical energy efficiency presented through Eq. (52). This is because Eq. (52) applies to a different general formula (Eq. (53)), where the endo- or exothermicity of the reaction determines the energy

efficiency. For example, for the same specific energy input, the energy efficiency drops significantly when the selectivity towards very stable products (e.g., H_2O during DRM) increases, as it lowers the endothermicity of the reaction. Therefore, it is no surprise that there is such a large difference when not all products are considered when deriving the reaction enthalpy (Eq. (52)* in Fig. 2), which is overestimated with 11.1% (equal to a relative deviation of 39.9%). Taking the reaction enthalpy for the "ideal" stoichiometry for DRM (Eq. (52)** in Fig. 2) described by reaction (R2), leads to an overestimated value of 3.5% (equal to a relative deviation of 12.6%). Finally, in the case of CO_2 hydrogenation, the same general trends are observed, with the chemical energy efficiency even reaching a negative value. More details on this can be found in the SI, Section S4.

From this numerical example, it is clear that a one-on-one comparison of reported energy efficiencies becomes challenging. Authors should always check the applied definitions and, if necessary, recalculate the values from other studies to allow for a fair comparison.

5.3 Critical evaluation of literature reports

In contrast to pure CO₂ splitting (Section 3.3), there is not much recent literature for plasma-based DRM and CO₂ hydrogenation where the effect of gas expansion or contraction is completely ignored and where the conversion is determined incorrectly. Hence, the paper of Pinhão et al. [20] had a good impact. Even though some authors explain the necessity of the flux ratio in their calculations incorrectly, also within our own group, they still do apply it in the correct way (e.g., Ref. [80]). Three different methods are described in literature to determine the flux ratio, i.e., using a standard component (e.g., Refs. [80–90]), a flow meter (e.g., Refs. [91–96]) and a stoichiometrically derived expression (e.g., Ref. [97]). We compare these three methods in detail in Section 6. Some authors define their formulas correctly (based on molar flow rates and not on fractions), but provide no details on the method used to determine the flux ratio (e.g., Refs. [98–104]), so we can only assume that they used one of these methods correctly. When the flux ratio is measured, the performance metrics, as described in Section 4, can be calculated. However, the example in Section 5.2 demonstrates that it can be difficult to interpret the results of various definitions. Therefore, we illustrate the lack of uniformity in this section with some examples from literature.

First is the energy cost, for which most definitions align with our general equation in Eq. (39). Sometimes it is defined relative to a product of interest, such as the syngas energy cost of Eq. (47) in Refs. [85, 88, 105], which is a valuable alternative. The same is true for the energy yield (e.g., a "methanol energy yield" is defined in Ref. [96]).

Second, as discussed in Sections 5.1.2 and 5.2, there is not one correct definition of energy efficiency, but a wide variety of them. For instance, Eq. (41) is used in Ref. [20], Eq. (49) in Ref. [82], Eq. (50) in Ref. [101], Eq. (51) in Refs. [84, 87, 89, 93, 97, 103] and Eq. (52) in Ref. [20]. However, there are some cases where an incorrect form of Eq. (52) is used. In Ref. [80], not all products are considered. While these authors motivate using their equation, i.e., excluding the hydrocarbons due to their negligible effect, they have a significant fraction of H₂O, which they exclude as well. However, it heavily alters the result, as shown in Section 5.2. In Refs. [86,102], multiple "ideal" reaction enthalpies are used. As explained in Sections 5.1.2 and 5.2, this is very often an incorrect expression. Indeed, in Ref. [86], other products are formed that are not described by the reaction equations corresponding to the used enthalpies. In Ref. [102], it seems that the by-products are not present in significant amounts, and then (and only then!) one can assume the ideal stoichiometry for the two main reactions to be valid. Nevertheless, the energy efficiency in this specific case is overestimated, since the reaction enthalpy is not describing the amount of energy per mole reactant. Rather, it describes the energy per 2 mole reactant, as it is 0.427 eV that is consumed when 1 mole CO₂ reacts with 1 mole H₂ towards CO and

 H_2O . When dividing by the energy cost in eV per 1 mole reactant converted, they do not refer to the same amounts of moles, and the resulting energy efficiency value should be divided by 2 to correct for this error.

Importantly, some authors express an energy efficiency in units of mmol kJ^{-1} (e.g., [86,91,93,95,98,99,101,103,104]). However, as mentioned in Sections 2.4 and 4.3, this should be called energy yield (see Eq. (17)). Alternatively, the energy yield can also be defined in units of g(H₂) kW h⁻¹, as in Ref. [92].

Notably, to compare energy efficiencies for DRM or CO₂ hydrogenation, one needs to carefully check whether the same formulas are used. Otherwise, the energy efficiency should be re-calculated with available raw data. In Refs. [56,93,101], the authors report a comparison of energy efficiencies from literature, but the values from literature are not the same as those reported, so we conclude the authors indeed re-calculated these values themselves, using the same formulas as used for their own energy efficiency. However, in Ref. [18], the authors refer, e.g., to values from Ref. [80], where different formulas were applied, hence this comparison is incorrect. Nevertheless, the impact on their conclusions remains limited because they also compared energy costs, which were calculated through the same formula. Due to the lack of a uniform definition for energy efficiency, we generally recommend comparing energy costs instead of energy efficiencies, as was also recommended in Ref. [9].

Third, the product selectivity and yield are reported more consistently in the field, with some exceptions, e.g., an incorrect definition of yield is used in Ref. [82] as explained in Eq. (58) (see Section 5.1). However, overall, literature reviews typically do not compare the yield in plasma reactors, but the conversion (e.g., Refs. [9,79]). Yet, plasma-based CO_2 conversion with a co-reactant is complex, and many products can be formed. It is therefore interesting to compare the yield of the desired product as well. Indeed, for the same energy cost, a high conversion is only meaningful when it is combined with a high selectivity toward the desired product. We will discuss this performance metric in more detail in our recommendations in Section 7.

We can conclude that although the effect of gas expansion and contraction is widely acknowledged in literature for plasma-based CO₂ conversion with a co-reactant, it remains challenging to interpret the variety of definitions, specifically for energy cost and energy efficiency. Again, transparency in data and calculations is key to facilitate an accurate comparison of literature reports, and we hope that this paper can provide insights to report more consistent and correct values for plasma technology.

6. Analysis of the available methods for determining the flux ratio

In the previous sections, we described the formulas for determining the different performance metrics for pure CO₂ conversion, DRM and CO₂ hydrogenation. The flux ratio is a fundamental part of these calculations. In addition, the stoichiometric equations, as presented in Section 2.1 for pure CO₂ splitting, are typically not applicable for more complex reactions, when by-products can be formed that cannot all be easily quantified or even detected. In that case, knowledge of the flux ratio is essential.

In Section 6.1, we describe the three main experimental methods to determine the flux ratio: (i) the stoichiometric relationships (when they are possible, i.e., as extension of Section 2.1), the use of (ii) a flow meter and of (iii) a standard component. All other performance metrics can be determined when the flux ratio is known, as described in Section 4. In literature, several researchers report one of these methods, as described in Section 5.3. However, to our knowledge, a detailed comparison between the different methods and specifically, the sensitivity of the performance metrics to these measurements, has not been reported. We will proceed with such a comparison in Section 6.2, supported by an experimental example for CO_2 splitting in a DBD reactor, and in Section 6.3, again supported by an experimental example for the conversion of CO_2 -CH₄-O₂ mixtures, in an atmospheric pressure glow discharge (APGD) reactor.

6.1 Description of the methods

6.1.1 Stoichiometric relationships

In Section 2.1, we discussed the stoichiometric relationships between the fractions of CO₂, CO and O₂ and the conversion for pure CO₂ splitting. Measuring only one of the fractions (y_{CO_2} , y_{CO} or y_{O_2}), or otherwise the flux ratio (α) (cf. Section 2.2), is sufficient to derive the conversion and each of the remaining parameters, as shown in the SI (Section S1). The advantage of this method is that the experimental setup can be simple. In theory, e.g., one accurate O₂-sensor is sufficient to determine all other parameters, but ideally, more than one parameter should be measured to validate the method. When the measurement is done accurately, the performance metrics derived from each measured parameter should all be the same (within the experimental error margin). To measure y_{CO_2} , y_{CO} and y_{O_2} , there are a number of techniques available. Dedicated sensors for each of these three components are widely available, and GC or FTIR spectrometers can be used to measure multiple components simultaneously.

Similar analytics are also applicable for DRM or CO_2 hydrogenation. However, with these reactions, stoichiometric relationships can only be derived if one is able to solve one of the three atom balance equations towards the flux ratio. The general formula for the atom balance b_A (where A can be C, H or O) is written as:

$$b_{A} = \frac{\alpha^{\text{fin}} \cdot \left(\sum_{i} (\mu_{i}^{A} \cdot y_{i}^{\text{out}}) + \sum_{j} (\mu_{j}^{A} \cdot y_{j}^{\text{out}})\right) + \alpha^{\text{init}} \cdot \sum_{i} (\mu_{i}^{A} \cdot y_{i}^{\text{out}})}{\sum_{i} (\mu_{i}^{A} \cdot y_{i}^{\text{in}})}$$
(59)

By definition, the atom balance should always be 1. In other words, solving Eq. (59) towards α^{fin} (and hence: using this method) is only possible if all reactants and products containing atom A, at inlet and outlet, are identified and their fraction quantified. Such an example will be discussed in more detail in Section 6.3.

6.1.2 Flow meter

A more direct measurement of the flux ratio can be done by means of a flow meter. The simplest example is a bubble flow meter, where the movement of a soap bubble across a certain distance is timed to derive the flow rate. By measuring this value before and during plasma reaction, at the same temperature and pressure, α is obtained through Eq. (6) in Section 2.2. In fact, any device that can directly or indirectly provide an accurate volumetric flow rate is suitable for this method.

6.1.3 Standard component

An indirect measurement of the flow rate can be performed by adding a standard component to the outlet stream of the reactor, as introduced by Pinhão et al. [20]. This standard component can be any gas (e.g., Ar, N_2) that does not react in the outflow. While the volumetric flow rate of the standard component does not change, its fraction will, but only due to the change in total flow rate. Please note that it is important for this method to introduce the standard component in the effluent gas and not in the feed gas. The addition of any gas to the inlet of the reactor will interfere with the plasma properties and influence the effective and total conversion, as well as the energy cost, energy yield and energy efficiency.

For pure CO_2 splitting, the same formulas for diluted CO_2 of Section 2.3 are valid to derive the conversion from the product fractions. Calculating the flux ratio, however, will not be the same when the standard is added to the outlet stream instead of as part of the feed gas. This is explained in more detail in the SI (Section S5.1).

The situation becomes more complex for reactions such as DRM or CO_2 hydrogenation, since the total gas flow rate and the fractions of the components change consecutively due to gas expansion, removal of liquid products and the addition of a standard, before finally reaching the analytical equipment. The simplest way to define the flux ratio is as the ratio of the measured fractions of the standard y_s^{in} and y_s^{out} :

$$\alpha_s^{\text{meas}} = \frac{y_s^{\text{in}}}{y_s^{\text{out}}} \tag{60}$$

Eq. (60) can then be used together with the measured input and output fractions to calculate all performance metrics presented in Section 4. However, there are multiple ways to express the flux ratio and the fractions that will lead to a correct determination of conversion, energy cost etc. For example, one can choose to work with the fractions in the gas stream before the standard is added, together with the flux ratio at that stage. Or, as explained in Section 4.3, one might need to derive the "initial" flux ratio (before condensation occurs). How this should be done is presented in detail in the SI (Section S5.2).

6.2 Example of pure CO₂ splitting in DBD plasma

We performed two separate series of experiments (one with a standard, one with a flow meter) with pure CO_2 flowing in a DBD plasma reactor. Experimental details can be found in the SI (Section S6). The performance metrics are calculated according to the formulas described in Section 2, and in the SI (Sections S1, S2 and S5.1).

The absolute CO_2 conversion and flux ratio in the first series of experiments are presented in Fig. 3(a), comparing the stoichiometrically derived values (based on the fraction of CO_2 , CO and O_2) with the values obtained using the standard component (N₂). The same parameters from the second series of experiments are presented in Fig. 3(b), comparing the stoichiometrically derived values with the values derived from the flow meter. To avoid confusion, we emphasize that the stoichiometric values (and

their error margins) in Fig. 3(a and b) are obtained through two separate series of experiments and are therefore not completely equal. The numerical data and results for all remaining performance metrics are given in the SI (Section S6).



Fig. 3. CO_2 conversion (left axis) and flux ratio (right axis) for pure CO_2 splitting in a DBD reactor. The results are determined by means of the stoichiometric formulas (based on the CO_2 , CO and O_2 fractions, first three bars), and the standard component (a) and bubble flow meter (b) (last bar in both cases). Error bars are added but some might be too small to be visible.

From both figures, it is clear that the CO_2 -, CO- and O_2 -based conversion and flux ratio are always very close to one another, almost always within each other's error margin. Hence, this data indicates that the stoichiometric relationships are suitable to derive the conversion with high accuracy. Notably, the results based on the N_2 standard and the flow meter are not very accurate, given that the error bar is significantly larger for these values. To explain this observation, we provide the relative errors on the fractions used to derive the conversion and flux ratio, as well as the relative error on the conversion

and flux ratio itself, in Table 5. The relative error is calculated similarly as was done in Eq. (37) in Section 3.2 for the conversion, now with the experimental absolute error used in the numerator.

Table 5. Relative errors on the output fractions, flux ratio and conversions, for both the experiments performed with the

| | | | 4 | | | | | | | |
|----------|---------|--------------|------------------------|---------------|----------------|----------------------|-----------|---------------|---------------|-------|
| | | | CO ₂ | | CO | | 02 | S | Standard | |
| 0.5%. | | | | | | | | | | |
| standard | added a | and the flow | meter used. | Note that the | relative error | on <i>B</i> , used i | n the exp | periments wit | h the standar | d, is |

| | 002 | 00 | 02 | Dunuara |
|------------------|-----------------|-------|-------|-------------------|
| y ^{out} | 0.4% | 0.3% | 0.6% | 1% |
| α | 0.2% | 0.09% | 0.09% | 2% |
| χ | 4% | 0.3% | 0.6% | 38% |
| | CO ₂ | CO | 02 | Flow meter |
| y ^{out} | 0.4% | 0.3% | 0.6% | - |
| α | 0.1% | 0.01% | 0.02% | 1% |
| X | 4% | 0.3% | 0.7% | 29% |

It is important to note that the relative error on the fractions, used to calculate the conversions and flux ratio, are all very comparable. After propagation of this error to the flux ratio, the majority of the errors decrease and all remain acceptable, with a maximum relative error of 2% on the flux ratio obtained through the fraction of the standard. However, when propagating these errors to the conversion, we see that the relative errors all increase significantly. The errors based on the measured fractions remain acceptable, but the ones based on the flow measurement propagate to a relative error of 38% and 29% with the standard and the flow meter, respectively. This proves that any small error on the flow rate and consequently flux ratio, will propagate to a much more significant error on the conversion, independent of the technique used and the reproducibility of the experiment. Hence, the conversions based on the measured CO₂, CO and O₂ fractions are in this case much more reliable.

Therefore, we conclude that using a standard component or a flow meter leads to more errors due to the sensitivity inherent to these methods. Instead, accurate diagnostics such as a GC should be used to determine all product fractions when possible, so that the more reliable stoichiometric values can be applied. They are easier to validate and obviously require less effort from the operator.

6.3 Example of DRM in the presence of O2 in APGD plasma

We again performed two separate series of experiments (one with a standard, one with a flow meter) similar to the experiments in the DBD reactor. Here, we use a mixture of CO_2 , CH_4 and O_2 (so-called oxidative CO_2 reforming of methane (OCRM)) in an APGD plasma reactor. Experimental details can be found in the SI (Section S7).

As explained in Section 6.1.1, to derive a stoichiometric relationship for DRM, one of the three atom balances needs to be solved to derive the final flux ratio. In this example, the APGD plasma is a socalled warm plasma, and under these conditions, by far the dominant product is syngas. High-value liquid products such as methanol can also be formed, but their fraction is often negligible. We used a cold trap between the reactor and the GC in these experiments, and the liquid fraction was analyzed after each experiment. Since the liquid consisted mainly of H_2O , with a fraction of 99.95%, the fraction of other components was negligible. In addition, a small amount of solid carbon was deposited in the experiments, specifically at the electrodes. However, this was less than a few milligrams, accumulated over time, so we conclude that its concentration is also negligible in the entire exhaust gas flow. Finally, all peaks on the chromatograms obtained with the GC were identified and quantified. This resulted in H_2O being the only significant product of which the fraction could not be measured. Hence, all C-based products could be identified, and therefore, the carbon balance equation (i.e., Eq. (52) of Section 6.1.1) can be re-arranged (considering that $b_c = 1$), to result in Eq. (61):

$$\alpha^{\text{fin}} = \frac{\sum_{i} (\mu_{i}^{\text{C}} \cdot y_{i}^{\text{in}})}{\sum_{i} (\mu_{i}^{\text{C}} \cdot y_{i}^{\text{out}}) + \sum_{j} (\mu_{j}^{\text{C}} \cdot y_{j}^{\text{out}})}$$
(61)

The formulas described in Section 4 were used to calculate the performance metrics. When the standard was added, the flux ratio as described in Section 6.1.3 was used, but also the other flux ratios were calculated, as described in the SI (Section S5.2).

The total conversion (i.e., the sum of the effective CO₂, CH₄ and O₂ conversions) and final flux ratio obtained with the first series of experiments are presented in Fig. 4(a), comparing the stoichiometrically derived values with the values derived by the standard component. The same parameters obtained with the second series of experiments are presented in Fig. 4(b), comparing the stoichiometrically derived values with the values obtained by the flow meter. To avoid confusion, we emphasize again that the stoichiometric values (and their error margins) in Fig. 4(a and b) are obtained through two separate series of experiments and are therefore not completely equal. The numerical data and results for all remaining performance metrics are given in the SI (Section S7).



Fig. 4. Total conversion and the effective CO_2 , CH_4 and O_2 conversions (left axis), and flux ratio (right axis) with OCRM in an APGD reactor. The results are determined by means of the stoichiometric formulas (based on the assumption of a perfect C-balance) (left bar), and the standard component (a) and bubble flow meter (b) (right bar in both cases).

Both the effective and total conversions obtained with the standard component and bubble flow meter are in good agreement with the values obtained by the stoichiometric formula. Indeed, the agreement, also in terms of error margins, is better than in Section 6.2, because in contrast to CO_2 splitting, the conversion here is much higher (i.e., 60%–70% total conversion). As a consequence, for the same absolute difference on the conversion, the relative difference becomes almost ten times smaller.

The difference between the flux ratios for the second series of experiments is negligible (0.01 ± 0.01), but for the first series of experiments, obtained with the standard component, the difference is a bit larger (i.e., 0.05 ± 0.01 , or a relative difference of around 4%–5%). For the values derived through the standard, it is also observed that the sum of all C-based selectivities is equal to $107\pm2\%$, and the sum of the C-based yields is higher than the C-based conversion. This is impossible because it implies a violation of the law of mass conservation. To explain this issue, we show the atom balances from the first series of experiments in Fig. 5(a), comparing the stoichiometrically derived values with the values obtained by using the standard component. The same is done for the values of the second series of

experiments in Fig. 5(b), comparing the stoichiometrically derived values with the values derived by the flow meter.



Fig. 5. Atom balance for C, H and O with OCRM in an APGD reactor. The results are determined by means of the stoichiometric formulas (based on assuming a perfect C-balance) (left bar), and the standard component (a) and bubble flow meter (b) (right bar in both cases).

Fig. 5 illustrates that all atom balances are close to 1 for the stoichiometric method (within their error margin). However, the C-balance obtained with the standard component, as shown in Fig. 5(a), clearly exceeds the theoretical maximum value of 1 (i.e., 1.04 ± 0.01). Since the fractions and the calibration of the different components are all the same, this deviation can only originate from an incorrect measurement of the flux ratio. While the effect on the conversion was nearly negligible (see Fig. 4a), the error of the flow measurements is now more apparent when looking at (the sum of) the product selectivities and the atom balances. Hence, results that would otherwise be inexplicable (like a violation of the law of mass conservation) can be traced back to a small error on the measurement of the flux ratio.

Based on these observations, we can again conclude that the stoichiometrically derived values seem to be the most reliable, similar to Section 6.2. Nevertheless, we want to stress again that it is often not

possible to apply the stoichiometric formulas for DRM (and similar reactions, like OCRM), because the composition of the liquid and solid fraction has to be known, and their original fraction in the gas mixture needs to be estimated at least.

Next, the bubble flow meter turned out to be more reliable in our experiments than the use of the standard component, although it is also less convenient to use. Not only does it require careful control of temperature and pressure, but it also requires manual action, prone to human error and limited to a certain maximum flow rate. An accurate digital flow meter is a better alternative, since it usually corrects for temperature and pressure effects, and yields a continuous reading of the flow rate. However, in our experience, a flow meter that provides the total volumetric flow without prior knowledge on the gas composition (as is the case at the reactor outlet) and that can be used for almost any gas mixture, is not easy to find. Moreover, it is not as affordable as the other methods described in this section.

The use of a standard component is more simple, because it can be analyzed together with the other components, can be very cheap and is easily introduced. However, it performs clearly the worst in this example. There are several plausible reasons: a small error on the MFC regulating the flow rate of the standard could already cause a small deviation on its measured fraction and the flux ratio, and propagate to a significant error on certain performance metrics such as the product selectivity. Another possibility is that the mixing of the standard with the rest of the gas mixture is not sufficient. We observed different results when introducing the standard at different positions (i.e., very close to the reactor outlet compared to more downstream), which could indicate a mixing issue. Finally, the way it is described by Pinhão et al. [20], there is no need to calibrate the analytics for the standard component itself. While this is true when the response of the analytic device on the standard component is a linear trend as function of its fraction, with the intercept at the *y*-axis equal to zero, we believe it is better to calibrate for this standard component as well. Not all analytics have a perfect linear response, and assuming so could have a significant impact on the flux ratio and the performance metrics.

Altogether, we advise checking the available methods for each new set of experiments. A comparison as in this section is relatively easy, and in our opinion, it is the best way to validate the chosen experimental method.

7. Recommendations and good practices

When reporting the performance of plasma-based CO_2 conversion, there are some important good practices to keep in mind. We recommend the excellent topical review by Alves et al. [106] on best practices in the field of low-temperature plasmas. Discussing challenges ranging from plasma diagnostics to computer models, the authors also address the importance of validation and reporting. Overall, they lay a clear foundation on how to improve communication, reproducibility and transparency within the field. In the following discussion, we present our own recommendations more specifically for measuring and reporting the performance for plasma-based CO_2 conversion, for pure CO_2 but also with co-reactant, such as in DRM and CO_2 hydrogenation.

In first instance, all details on the experimental setup should be provided. For the power, it is important to note the difference between (i) the plasma power, which is the true power dissipated in the plasma, (ii) the applied power, which is the power delivered by the PSU, and (iii) the plug power, which is the overall power consumed by the PSU during plasma operation. The applied power is equal to the plasma power plus the power related to the losses in the electronic circuit, while the plug power also accounts for the efficiency of the PSU itself. The relation between the plug power and the plasma power is given by the so-called plug-to-plasma power efficiency. Typically, the measured plasma power is used for calculating the energy efficiency and energy cost, as it reflects the efficiency of the plasma process itself. However, in an industrial context, the plug-to-plasma power efficiency should be accounted for, and thus, the applied power should be used to calculate the overall energy efficiency and energy cost [107]. In addition, the reported power should also include the duty cycle when a pulsed power is applied [108]. Special care should be taken to describe the electrical components (i.e., voltage probe, current sensor, ...) and calculations that were used to obtain the plasma power.

Moreover, when measuring the outlet gas composition in a wide range of conditions, all analytical equipment should be carefully calibrated, starting with the MFCs, because small deviations can lead to large errors for determining the flux ratio. Regular calibrations are needed according to the manufacturer's recommendations, but at least before every new set of experiments.

Furthermore, the sampling position in the setup should be specified, especially for warm plasmas. Most experiments measure the outlet gas downstream from the reactor, where possible recombination reactions (e.g., $CO+O_2$ reacting back into CO_2) have reached equilibrium with the forward (splitting) reactions, and the gas composition has reached a steady state. The conversions downstream are thus typically lower than compared to sampling right after the plasma reactor and one should pay special attention that the sampling position is the same when comparing different experiments. In general, all details of the experiment should be described for transparency and repeatability. Although this may seem evident, not all authors in the field provide the proper descriptions.

When all the experimental details are specified, presenting the data in a clear and transparent way is equally important. For each presented data point, all experimental conditions should be specified: the flow rates of both the input gases and dilution agents, as well as the power, and the fractions of the output gases, together with the flux ratio, to derive the flow rates of the output gases. Ideally, the raw data should be open access to allow other researchers to reanalyze the data at a later stage and to comply with the FAIR principles [109]. How to present data with statistical relevance and other detailed recommendations for plasma research in general, can be found in the review by Alves et al. [106].

Last but not least, when all experimental details are provided, the correct formulas should be used for the calculation of the performance (i.e., conversion, yields, selectivity, energy cost and energy

efficiency), and they should be clearly reported. When doing a comparison of different setups, we recommend to pay special attention to the product yield and the energy cost or yield. The product yield of the desired products is a very informative parameter, because it is the product of both conversion and selectivity. Secondly, energy cost or yield should be compared, rather than energy efficiency, due to the lack of a uniform definition for the latter.

We hope that this section, together with the correct formulas presented in previous sections, can provide a guideline for more consistent calculations, and can stimulate a correct comparison of different plasma reactors and conditions described in literature.

8. Conclusions

Over the past decades, plasma technology has gained increasing interest as possible CCU technology. However, the formulas used in literature to calculate the performance metrics are often inconsistent, and sometimes even incorrect, leading to serious deviations from the real performance. This makes it challenging to compare the performance of plasma processes between different experiments (e.g., different conditions and plasma reactors) from different research groups, and hinders the development of this technology on both a fundamental and more applied level. In this critical review, we summarized the correct formulas for the performance metrics and highlighted common inconsistencies and errors, alongside general points of attention and recommendations.

We presented stoichiometrically derived expressions to simply and correctly calculate the conversion for pure CO_2 splitting, which are rarely used in literature so far, alongside with the correct definitions for energy cost, energy yield and energy efficiency. Furthermore, we showed that incorrect definitions for conversion are common in literature, due to neglecting the change in volumetric flow rate, confirming earlier findings [20]. This leads to relative errors on the conversion as high as 40%, which we demonstrated with a numerical example.

Next, we summarized the correct definitions for the performance metrics of CO₂ conversion with a coreactant, specifically for the dry reforming of methane and CO₂ hydrogenation. We showed that a wide variety of definitions exists for energy cost and especially energy efficiency. As a consequence, a direct comparison of energy efficiencies obtained with different reactors is complicated, which we demonstrated through a numerical example. Therefore, when comparing different studies for CO₂ mixtures, we recommend comparing energy cost or energy yield rather than energy efficiency, due to the inconsistent definitions for the latter, as was also recommended in Ref. [9]. Apart from the energy cost, the product yield is the second most informative performance metric in case of CO₂ conversion with a co-reactant, and more suitable than the conversion for a comparison between processes, because it also includes the selectivity to the desired products.

For CO₂ mixtures with a co-reactant, taking into account the flux ratio is crucial. It should be determined by means of a direct measurement (e.g., flow meter) or an indirect measurement (e.g., adding a standard component in the outlet stream of the reactor), and in some cases it can be derived stoichiometrically. We presented a detailed comparison of these methods, supported by two experimental examples, and investigated the experimental error on the conversion and the flux ratio. We found that for the same error, the uncertainty on the performance metrics can propagate to a much larger extent when using one method compared to another, and it is an important aspect that needs to be considered prior to any series of experiments. Therefore, we advise to always do such a simple comparison, as was also performed by Refs. [42,70].

Finally, we provided some general recommendations. Specifically, we advise to always report essential parameters, such as the plug-to-plasma power efficiency, all experimental input conditions, the flux ratio, the fractions of the output gases and the formulas used to calculate all performance metrics.

We can conclude that besides the clear importance of reporting all experimental details and applied formulas, more consistent formulas are essential to correctly analyze the performance in the field of plasma-based CO_2 conversion. We hope that this paper can provide a guideline for authors and facilitate objective and correct comparisons of different reactors, to stimulate further development of plasma technology and contribute to a more sustainable future.

Acknowledgments

We acknowledge financial support from the Fund for Scientific Research (FWO) Flanders (Grant ID 110221N), the European Research Council (ERC) under the European Union's Horizon 2020 Research and Innovation Program (grant agreement No 810182 - SCOPE ERC Synergy project) and the Methusalem funding of the University of Antwerp. We acknowledge the icons from the graphical abstract made by dDara, geotatah, Spashicons and Freepik on www.flaticon.com. We also thank Stein Maerivoet, Joachim Slaets, Elizabeth Mercer, Colín Ó'Modráin, Joran Van Turnhout, Pepijn Heirman, dr. Yury Gorbanev, dr. Fanny Girard-Sahun and dr. Sean Kelly for the interesting discussions and feedback.

References

[1] S. Valluri, V. Claremboux, S. Kawatra, J. Environ. Sci. 113 (2022) 322-344.

[2] C. Hepburn, E. Adlen, J. Beddington, E.A. Carter, S. Fuss, N. Mac Dowell, J.C. Minx, P. Smith, C.K. Williams, Nature 575 (2019) 87-97.

[3] A. Al-Mamoori, A. Krishnamurthy, A.A. Rownaghi, F. Rezaei, Energy Technol. 5 (2017) 834-849.

[4] H. Mikulčić, I. Ridjan Skov, D.F. Dominković, S.R. Wan Alwi, Z.A. Manan, R. Tan, N. Duić, S.N. Hidayah Mohamad, X. Wang, Renewable Sustainable Energy Rev. 114 (2019) 109338.

[5] E.I. Koytsoumpa, C. Bergins, E. Kakaras, J. Supercrit. Fluids 132 (2018) 3-16.

[6] S.M. Jarvis, S. Samsatli, Renewable Sustainable Energy Rev. 85 (2018) 46-68.

[7] M.A. Sabri, S. Al Jitan, D. Bahamon, L.F. Vega, G. Palmisano, Sci. Total Environ. 790 (2021) 148081.

[8] T.A. Saleh, RSC Adv. 12 (2022) 23869-23888.

[9] R. Snoeckx, A. Bogaerts, Chem. Soc. Rev. 46 (2017) 5805-5863.

[10] G. Chen, R. Snyders, N. Britun, J. CO2 Util. 49 (2021) 101557.

[11] F. Jardali, S. Van Alphen, J. Creel, H.A. Eshtehardi, M. Axelsson, R. Ingels, R. Snyders, A. Bogaerts, Green Chemistry 23 (2021) 1748-1757.

[12] S. Kelly, A. Bogaerts, Joule 5 (2021) 3006-3030.

[13] K.H. Rouwenhorst, Y. Engelmann, K. van't Veer, R.S. Postma, A. Bogaerts, L. Lefferts, Green chemistry 22 (2020) 6258-6287.

[14] K.H. Rouwenhorst, F. Jardali, A. Bogaerts, L. Lefferts, Energy Environ. Sci. 14 (2021) 2520-2534.

[15] E. Delikonstantis, F. Cameli, M. Scapinello, V. Rosa, K.M. Van Geem, G.D. Stefanidis, Curr. Opin. Chem. Eng. 38 (2022) 100857.

[16] A. George, B. Shen, M. Craven, Y. Wang, D. Kang, C. Wu, X. Tu, Renewable Sustainable Energy Rev. 135 (2021) 109702.

[17] M.Y. Ong, S. Nomanbhay, F. Kusumo, P.L. Show, J. Clean. Prod. 336 (2022) 130447.

[18] J.-L. Liu, X. Wang, X.-S. Li, B. Likozar, A.-M. Zhu, J. Phys. D: Appl. Phys. 53 (2020) 253001.

[19] Y. Vadikkeettil, Y. Subramaniam, R. Murugan, P.V. Ananthapadmanabhan, J. Mostaghimi, L. Pershin, C. Batiot-Dupeyrat, Y. Kobayashi, Renewable Sustainable Energy Rev. 161 (2022) 112343.

[20] N. Dinhão, A. Moura, J.P. Branco, J. Novos, Int. J. Hydrogon Energy 41 (2016) 0245 0255

[20] N. Pinhão, A. Moura, J.B. Branco, J. Neves, Int. J. Hydrogen Energy 41 (2016) 9245-9255.

[21] F. Peeters, T. Butterworth, Electrical Diagnostics of Dielectric Barrier Discharges, in: N. Anton, C. Zhiqiang (Eds.) Atmospheric Pressure Plasma, IntechOpen, Rijeka, 2018, pp. Ch. 2.

[22] M. Ramakers, G. Trenchev, S. Heijkers, W. Wang, A. Bogaerts, ChemSusChem 10 (2017) 2642-2652.

[23] S. Kelly, C. Verheyen, A. Cowley, A. Bogaerts, Chem 8 (2022) 2797-2816.

[24] J.G. Calvert, 62 (1990) 2167-2219.

[25] ISO, (1994) https://www.iso.org/standard/18855.html (Accessed 06 July 2023)

[26] M.J. Berger, J.S. Coursey, M.A. Zucker, J. Chang, NIST (2017) https://www.nist.gov/pml/stopping-power-range-tables-electrons-protons-and-helium-ions (Accessed 06 July 2023).

[27] J.D. Wright, A.N. Johnson, M.R. Moldover, J. Res. Natl. Inst. Stand. Technol. 108 (2003) 21-47.

[28] P. Wu, X. Li, N. Ullah, Z. Li, Mol. Catal. 499 (2021) 111304.

- [29] K. Wiegers, A. Schulz, M. Walker, G.E.M. Tovar, Chem. Ing. Tech. 94 (2022) 299-308.
- [30] R. Snoeckx, S. Heijkers, K. Van Wesenbeeck, S. Lenaerts, A. Bogaerts, Energy Environ. Sci. 9 (2016) 999-1011.
- [31] G. Trenchev, A. Bogaerts, J. CO2 Util. 39 (2020) 101152.
- [32] Y. Uytdenhouwen, S. Van Alphen, I. Michielsen, V. Meynen, P. Cool, A. Bogaerts, Chem. Eng. J. 348 (2018) 557-568.
- [33] V. Ivanov, T. Paunska, S. Lazarova, A. Bogaerts, S. Kolev, J. CO2 Util. 67 (2023) 102300.
- [34] G. Niu, Y. Qin, W. Li, Y. Duan, Plasma Chem. Plasma Process. 39 (2019) 809-824.
- [35] B. Raja, R. Sarathi, R. Vinu, Energy Technol. 8 (2020) 2000535.
- [36] A.J. Wolf, F.J.J. Peeters, P.W.C. Groen, W.A. Bongers, M.C.M. van de Sanden, J. Phys. Chem. C 124 (2020) 16806-16819.
- [37] H. Sun, Z. Chen, J. Chen, H. Long, Y. Wu, W. Zhou, J. Phys. D: Appl. Phys. 54 (2021) 495203.
- [38] S. Renninger, M. Lambarth, K.P. Birke, J. CO2 Util. 42 (2020) 101322.
- [39] F. D'Isa, E. Carbone, A. Hecimovic, U. Fantz, Plasma Sources Sci. Technol. 29 (2020) 105009.
- [40] F. Girard-Sahun, O. Biondo, G. Trenchev, G.J. van Rooij, A. Bogaerts, Chem. Eng. J. 442 (2022) 136268.
- [41] A. Hecimovic, F.A. D'Isa, E. Carbone, U. Fantz, J. CO2 Util. 57 (2022) 101870.
- [42] J. Huang, H. Zhang, Q. Tan, L. Li, R. Xu, Z. Xu, X. Li, J. CO2 Util. 45 (2021) 101429.
- [43] D.C.M. van den Bekerom, J.M.P. Linares, T. Verreycken, E.M. van Veldhuizen, S. Nijdam, G. Berden,
- W.A. Bongers, M.C.M. van de Sanden, G.J. van Rooij, Plasma Sources Sci. Technol. 28 (2019) 055015.
- [44] H. Zhang, L. Li, R. Xu, J. Huang, N. Wang, X. Li, X. Tu, Waste Dispos. Sustain. Energy 2 (2020) 139-150.
- [45] J. Li, X. Zhang, J. Shen, T. Ran, P. Chen, Y. Yin, J. CO2 Util. 21 (2017) 72-76.
- [46] M. Zhu, S. Hu, F. Wu, H. Ma, S. Xie, C. Zhang, J. Phys. D: Appl. Phys. 55 (2022) 225207.
- [47] I. Belov, V. Vermeiren, S. Paulussen, A. Bogaerts, J. CO2 Util. 24 (2018) 386-397.
- [48] A. Barkhordari, S. Karimian, A. Rodero, D.A. Krawczyk, S.I. Mirzaei, A. Falahat, Appl. Sci. 11 (2021) 10047.
- [49] J.O. Pou, E. Estopañán, J. Fernandez-Garcia, R. Gonzalez-Olmos, Processes 10 (2022) 1851.
- [50] G. Trenchev, A. Nikiforov, W. Wang, S. Kolev, A. Bogaerts, Chem. Eng. J. 362 (2019) 830-841.
- [51] N. Britun, T. Silva, G. Chen, T. Godfroid, J. van der Mullen, R. Snyders, J. Phys. D: Appl. Phys. 51 (2018) 144002.
- [52] S.C.L. Vervloedt, M. Budde, R. Engeln, Plasma Sources Sci. Technol. 32 (2023) 015004.
- [53] H. Kim, S. Song, C.P. Tom, F. Xie, J. CO2 Util. 37 (2020) 240-247.
- [54] D. Zhang, Q. Huang, E.J. Devid, E. Schuler, N.R. Shiju, G. Rothenberg, G. van Rooij, R. Yang, K. Liu, A.W. Kleyn, J. Phys. Chem. C 122 (2018) 19338-19347.
- [55] O. Taylan, H. Berberoglu, Plasma Sources Sci. Technol. 24 (2015) 015006.
- [56] D. Mei, X. Tu, ChemPhysChem 18 (2017) 3253-3259.
- [57] C. Guoxing, B. Nikolay, G. Thomas, D.O. Marie-Paule, S. Rony, Role of Plasma Catalysis in the Microwave Plasma-Assisted Conversion of CO2, IntechOpen, Rijeka, 2017.
- [58] W. Bongers, H. Bouwmeester, B. Wolf, F. Peeters, S. Welzel, D. van den Bekerom, N. den Harder, A. Goede, M. Graswinckel, P.W. Groen, Plasma Process. Polym. 14 (2017) 1600126.
- [59] M. Umamaheswara Rao, K.V.S.S. Bhargavi, P. Chawdhury, D. Ray, S.R.K. Vanjari, C. Subrahmanyam, Chem. Eng. Sci. 267 (2023) 118376.
- [60] M.R. Jahanbakhsh, H. Taghvaei, O. Khalifeh, M. Ghanbari, M.R. Rahimpour, Energy & Fuels 34 (2020) 14321-14332.
- [61] A.M. Banerjee, J. Billinger, K.J. Nordheden, F.J.J. Peeters, Journal of Vacuum Science & Technology A 36 (2018) 04F403.
- [62] L. Wang, X. Du, Y. Yi, H. Wang, M. Gul, Y. Zhu, X. Tu, Chem. Commun. 56 (2020) 14801-14804.
- [63] B. Wang, X. Li, X. Wang, B. Zhang, J. Environ. Chem. Eng. 9 (2021) 106370.
- [64] Y. Qin, G. Niu, X. Wang, D. Luo, Y. Duan, Chem. Phys. 538 (2020) 110913.
- [65] K. Zhang, A.P. Harvey, Chem. Eng. J. 405 (2021) 126625.
- [66] C. Montesano, S. Quercetti, L.M. Martini, G. Dilecce, P. Tosi, J. CO2 Util. 39 (2020) 101157.

- [67] K. Zhang, G. Zhang, X. Liu, A.N. Phan, K. Luo, Ind. Eng. Chem. Res. 56 (2017) 3204-3216.
- [68] P. Liu, X. Liu, J. Shen, Y. Yin, T. Yang, Q. Huang, D. Auerbach, A.W. Kleiyn, Plasma Sci. Technol. 21 (2019) 012001.
- [69] Y. Wu, S.-Z. Li, Y.-L. Niu, H. Yan, D. Yang, J. Zhang, J. Phys. D: Appl. Phys. 56 (2022) 065201.
- [70] H. Zhang, Q. Tan, Q. Huang, K. Wang, X. Tu, X. Zhao, C. Wu, J. Yan, X. Li, ACS Sustain. Chem. Eng. 10 (2022) 7712-7725.
- [71] H. Taghvaei, E. Pirzadeh, M. Jahanbakhsh, O. Khalifeh, M.R. Rahimpour, J. CO2 Util. 44 (2021) 101398.
- [72] S. Xu, J.C. Whitehead, P.A. Martin, Chem. Eng. J. 327 (2017) 764-773.
- [73] D. Ray, R. Saha, S. Ch., Catalysts 7 (2017) 244.
- [74] T. Wang, H. Liu, X. Xiong, X. Feng, IOP Conf. Ser.: Earth Environ 52 (2017) 012100.
- [75] M. Ramakers, I. Michielsen, R. Aerts, V. Meynen, A. Bogaerts, Plasma Process. Polym. 12 (2015) 755-763.
- [76] T. Wang, 15 The gas and steam turbines and combined cycle in IGCC systems, in: T. Wang, G. Stiegel (Eds.) Integrated Gasification Combined Cycle (IGCC) Technologies, Woodhead Publishing, 2017, pp. 497-640.
- [77] L.R. Winter, J.G. Chen, Joule 5 (2021) 300-315.
- [78] V. Hessel, A. Anastasopoulou, Q. Wang, G. Kolb, J. Lang, Catal. Today 211 (2013) 9-28.
- [79] B. Wanten, S. Maerivoet, C. Vantomme, J. Slaets, G. Trenchev, A. Bogaerts, J. CO2 Util. 56 (2022) 101869.
- [80] E. Cleiren, S. Heijkers, M. Ramakers, A. Bogaerts, ChemSusChem 10 (2017) 4025-4036.
- [81] J. Slaets, M. Aghaei, S. Ceulemans, S. Van Alphen, A. Bogaerts, Green Chemistry 22 (2020) 1366-1377.
- [82] R. Snoeckx, W. Wang, X. Zhang, M.S. Cha, A. Bogaerts, Sci. Rep. 8 (2018) 15929.
- [83] B. Zhu, X.-S. Li, J.-L. Liu, X. Zhu, A.-M. Zhu, Chem. Eng. J. 264 (2015) 445-452.
- [84] K. Li, J.-L. Liu, X.-S. Li, X.-B. Zhu, A.-M. Zhu, Catal. Today 256 (2015) 96-101.
- [85] W.-C. Chung, M.-B. Chang, Energy Convers. Manag. 124 (2016) 305-314.
- [86] J.A. Andersen, J.M. Christensen, M. Østberg, A. Bogaerts, A.D. Jensen, Chem. Eng. J. 397 (2020) 125519.
- [87] C. Montesano, M. Faedda, L.M. Martini, G. Dilecce, P. Tosi, J. CO2 Util. 49 (2021) 101556.
- [88] D. Lašič Jurković, J.-L. Liu, A. Pohar, B. Likozar, Catal. Today 362 (2021) 11-21.
- [89] H. Kwon, T. Kim, S. Song, J. CO2 Util. 70 (2023) 102448.
- [90] F. Zhang, X. Zhang, Z. Song, X. Li, X. Zhao, J. Sun, Y. Mao, X. Wang, W. Wang, Fuel 331 (2023) 125914.
- [91] V. Shapoval, E. Marotta, Plasma Process. Polym. 12 (2015) 808-816.
- [92] S.M. Chun, Y.C. Hong, D.H. Choi, J. CO2 Util. 19 (2017) 221-229.
- [93] M. Biset-Peiró, R. Mey, J. Guilera, T. Andreu, Chem. Eng. J. 393 (2020) 124786.
- [94] R.B. Raja, A.C. Halageri, R. Sankar, R. Sarathi, R. Vinu, Energies 16 (2023) 1823.
- [95] M. Ronda-Lloret, Y. Wang, P. Oulego, G. Rothenberg, X. Tu, N.R. Shiju, ACS Sustain. Chem. Eng. 8 (2020) 17397-17407.
- [96] L. Wang, Y. Yi, H. Guo, X. Tu, ACS Catal. 8 (2018) 90-100.
- [97] H. Sun, J. Lee, M.S. Bak, J. CO2 Util. 46 (2021) 101464.
- [98] X. Tu, H.J. Gallon, M.V. Twigg, P.A. Gorry, J.C. Whitehead, J. Phys. D: Appl. Phys. 44 (2011) 274007.
- [99] N. Lu, D. Sun, Y. Xia, K. Shang, B. Wang, N. Jiang, J. Li, Y. Wu, Int. J. Hydrogen Energy 43 (2018) 13098-13109.
- [100] R. Dębek, F. Azzolina-Jury, A. Travert, F. Maugé, F. Thibault-Starzyk, Catal. Today 337 (2019) 182-194.
- [101] Y. Zeng, G. Chen, Q. Bai, L. Wang, R. Wu, X. Tu, Int. J. Hydrogen Energy 48 (2023) 6192-6203.
- [102] J.F. de la Fuente, S.H. Moreno, A.I. Stankiewicz, G.D. Stefanidis, Int. J. Hydrogen Energy 41 (2016) 21067-21077.
- [103] Y. Diao, H. Wang, B. Chen, X. Zhang, C. Shi, Appl. Catal., B 330 (2023) 122573.

[104] H. Wang, Y. Yang, Z. Li, X. Kong, P. Martin, G. Cui, R. Wang, Int. J. Hydrogen Energy 48 (2023) 8921-8931.

[105] K. Li, J.-L. Liu, X.-S. Li, X. Zhu, A.-M. Zhu, Chem. Eng. J. 288 (2016) 671-679.

[106] L.L. Alves, M.M. Becker, J. van Dijk, T. Gans, D.B. Go, K. Stapelmann, J. Tennyson, M.M. Turner, M.J. Kushner, Plasma Sources Sci. Technol. 32 (2023) 023001.

[107] I. Tsonev, C. O'Modhrain, A. Bogaerts, Y. Gorbanev, ACS Sustain. Chem. Eng. 11 (2023) 1888-1897.

[108] E. Vervloessem, Y. Gorbanev, A. Nikiforov, N. De Geyter, A. Bogaerts, Green Chemistry 24 (2022) 916-929.

[109] M.D. Wilkinson, M. Dumontier, I.J. Aalbersberg, G. Appleton, M. Axton, A. Baak, N. Blomberg, J.-W. Boiten, L.B. da Silva Santos, P.E. Bourne, J. Bouwman, A.J. Brookes, T. Clark, M. Crosas, I. Dillo, O. Dumon, S. Edmunds, C.T. Evelo, R. Finkers, A. Gonzalez-Beltran, A.J.G. Gray, P. Groth, C. Goble, J.S. Grethe, J. Heringa, P.A.C. 't Hoen, R. Hooft, T. Kuhn, R. Kok, J. Kok, S.J. Lusher, M.E. Martone, A. Mons, A.L. Packer, B. Persson, P. Rocca-Serra, M. Roos, R. van Schaik, S.-A. Sansone, E. Schultes, T. Sengstag, T. Slater, G. Strawn, M.A. Swertz, M. Thompson, J. van der Lei, E. van Mulligen, J. Velterop, A. Waagmeester, P. Wittenburg, K. Wolstencroft, J. Zhao, B. Mons, Sci. Data 3 (2016) 160018.

Graphic abstract

This review aims to provide a comprehensive guideline for authors to encourage more consistent calculations in plasma-based CO2 conversion and stimulate the further development of this technology.



Supporting information: Plasma-based CO₂ conversion: How to correctly analyze the performance?

Bart Wanten^{*},¹ Rani Vertongen^{*},¹ Robin De Meyer^{1,2} and Annemie Bogaerts¹

¹ Research group PLASMANT, Department of Chemistry, University of Antwerp, Universiteitsplein 1, 2610 Antwerp, Belgium.

² Research group EMAT, Department of Physics, University of Antwerp, Groenenborgerlaan 171, 2020 Antwerp, Belgium

Corresponding authors:

M. Sc Bart Wanten, email: bart.wanten@uantwerpen.be

Prof. Dr. Annemie Bogaerts, email: annemie.bogaerts@uantwerpen.be

* shared first author

S1. Formulas for pure CO₂ conversion: Relationships between conversion, CO₂/CO/O₂ fractions and flux ratio

The equations in Sections 2.1 and 2.2 in the main paper give the basic outline of the conversion calculation based on the fractions of CO_2 , CO and O_2 (Section 2.1) as well as the relationship with the flux ratio (Section 2.2). Depending on which information is available in the experiment, the other values can be deduced, as summarized in Table S.1. For example, when we measure the CO_2 output fraction $y_{CO_2}^{out}$, we can obtain the conversion, by rearranging Eq. (1) in the main paper to Eq. (4) as follows:

$$y_{\text{CO}_2}^{\text{out}} = \frac{1-\chi}{1+\frac{\chi}{2}}$$
$$y_{\text{CO}_2}^{\text{out}} \cdot \left(1+\frac{\chi}{2}\right) = 1-\chi$$
$$y_{\text{CO}_2}^{\text{out}} + \frac{y_{\text{CO}_2}^{\text{out}} \cdot \chi}{2} = 1-\chi$$
$$\chi + \frac{y_{\text{CO}_2}^{\text{out}} \cdot \chi}{2} = 1-y_{\text{CO}_2}^{\text{out}}$$
$$\chi \cdot \left(1+\frac{y_{\text{CO}_2}^{\text{out}}}{2}\right) = 1-y_{\text{CO}_2}^{\text{out}}$$
$$\chi = \frac{1-y_{\text{CO}_2}^{\text{out}}}{1+\frac{y_{\text{CO}_2}^{\text{out}}}{2}}$$

Table S.1. Overview of formulas for pure CO₂ conversion: How to obtain the conversion, flux ratio, or output fraction of CO₂, CO or O₂, when one of the other values is known.

| Known → Unknown ↓ | X | α | $\mathcal{Y}_{\mathrm{CO}_2}^{\mathrm{out}}$ | y _{co} | $y_{0_2}^{out}$ |
|--------------------------|---|-------------------------------------|--|--|---|
| χ = | χ | $2 \cdot \alpha - 2$ | $\frac{1 - y_{\text{CO}_2}^{\text{out}}}{1 + \frac{y_{\text{CO}_2}^{\text{out}}}{2}}$ | $\frac{y_{\rm CO}^{\rm out}}{1 - \frac{y_{\rm CO}^{\rm out}}{2}}$ | $\frac{2 \cdot y_{O_2}^{\text{out}}}{1 - y_{O_2}^{\text{out}}}$ |
| α = | $1+\frac{\chi}{2}$ | α | $\frac{3}{2 + y_{CO_2}^{out}}$ | $\frac{2}{2 - y_{\rm CO}^{\rm out}}$ | $\frac{1}{1-y_{0_2}^{\text{out}}}$ |
| $y_{CO_2}^{out} =$ | $\frac{1-\chi}{1+\frac{\chi}{2}}$ | $\frac{3-2\cdot\alpha}{\alpha}$ | $y_{\rm CO_2}^{\rm out}$ | $\frac{(y_{\text{CO}}^{\text{out}} - 2) \cdot (3 \cdot y_{\text{CO}}^{\text{out}} - 2)}{4 - 2 \cdot y_{\text{CO}}^{\text{out}}}$ | $1 - 3 \cdot y_{0_2}^{\text{out}}$ |
| $y_{\rm CO}^{\rm out} =$ | $\frac{\chi}{1+\frac{\chi}{2}}$ | $\frac{2 \cdot \alpha - 2}{\alpha}$ | $\frac{-(y_{CO_2}^{out}+2)\cdot(y_{CO_2}^{out}-1)}{3+\frac{3\cdot y_{CO_2}^{out}}{2}}$ | $y_{\rm CO}^{\rm out}$ | $2 \cdot y_{0_2}^{out}$ |
| $y_{0_2}^{out} =$ | $\frac{\frac{\chi}{2}}{1+\frac{\chi}{2}}$ | $\frac{\alpha-1}{\alpha}$ | $\frac{1 - y_{\rm CO_2}^{\rm out}}{3}$ | $\frac{y_{\rm CO}^{\rm out}}{2}$ | $y_{0_2}^{out}$ |

S2. Formulas for CO₂ conversion with a diluting agent

When adding a diluting agent to the feed gas, the dilution effect needs to be taken into account when calculating the performance metrics. When applying the same approach as described in Section 2.1 in the main paper, we obtain Table S.2. Note that these values are the flow rates of each species relative to the input CO_2 flow rate.

Table S.2. Reaction equation for pure CO_2 conversion, expressed in flow rates relative to the CO_2 flow rate at the reactor inlet, in the presence of a diluting agent.

| Reaction | $CO_2 \rightarrow$ | СО | 02 | Diluting agent |
|----------|--------------------|----|----------|----------------|
| in | 1 | 0 | 0 | β |
| out | $1-\chi$ | χ | $\chi/2$ | β |

When expressing the fraction of CO₂ at the outlet, we obtain the following expression:

$$y_{\rm CO_2}^{\rm out} = \frac{\dot{n}_{\rm CO_2}^{\rm out}}{\dot{n}_{\rm tot}^{\rm out}} = \frac{\dot{n}_{\rm CO_2}^{\rm out}/\dot{n}_{\rm CO_2}^{\rm in}}{\dot{n}_{\rm tot}^{\rm out}/\dot{n}_{\rm CO_2}^{\rm in}} = \frac{1-\chi}{(1-\chi)+\chi+\frac{\chi}{2}+\beta} = \frac{1-\chi}{1+\frac{\chi}{2}+\beta}$$
(5.1)

Rearranging this equation to express conversion as function of the CO_2 output fraction is done as follows:

$$y_{\text{CO}_2}^{\text{out}} = \frac{1-\chi}{1+\frac{\chi}{2}+\beta}$$
$$y_{\text{CO}_2}^{\text{out}} \cdot \left(1+\frac{\chi}{2}+\beta\right) = 1-\chi$$
$$y_{\text{CO}_2}^{\text{out}} + \frac{y_{\text{CO}_2}^{\text{out}}}{2} \cdot \chi + y_{\text{CO}_2}^{\text{out}} \cdot \beta = 1-\chi$$
$$\chi + \frac{y_{\text{CO}_2}^{\text{out}}}{2} \cdot \chi = 1-y_{\text{CO}_2}^{\text{out}} - y_{\text{CO}_2}^{\text{out}} \cdot \beta$$
$$\left(1+\frac{y_{\text{CO}_2}^{\text{out}}}{2}\right) \cdot \chi = 1-(1+\beta) \cdot y_{\text{CO}_2}^{\text{out}}$$
$$\chi = \frac{1-(1+\beta) \cdot y_{\text{CO}_2}^{\text{out}}}{1+\frac{y_{\text{CO}_2}^{\text{out}}}{2}}$$

This is the same equation as presented by Eq. (11) in the main paper. The relationships between the conversion, the flux ratio and the CO_2 , CO and O_2 fraction, as well as the fraction of the diluting agent, are presented in Table S.3.

| Known → Unknown ↓ | X | α | $\mathcal{Y}^{out}_{CO_2}$ | y _{co} ^{out} | $y_{0_2}^{out}$ | $y_s^{\rm out}$ |
|--------------------------|---|---|---|--|---|---|
| x = | X | $(2 \cdot \alpha - 2) \cdot (1 + \beta)$ | $\frac{1 - (1 + \beta) \cdot y_{\text{CO}_2}^{\text{out}}}{1 + \frac{y_{\text{CO}_2}^{\text{out}}}{2}}$ | $\frac{(1+\beta) \cdot y_{\rm CO}^{\rm out}}{1-\frac{y_{\rm CO}^{\rm out}}{2}}$ | $\frac{2 \cdot (1+\beta) \cdot y_{0_2}^{\text{out}}}{1-y_{0_2}^{\text{out}}}$ | $\frac{2 \cdot \beta - 2 \cdot (1 + \beta) \cdot y_s^{\text{out}}}{y_s^{\text{out}}}$ |
| α = | $\frac{1+\frac{\chi}{2}+\beta}{1+\beta}$ | α | $1 + \frac{1 - (1 + \beta) \cdot y_{\text{CO}_2}^{\text{out}}}{2 \cdot (1 + \beta) \cdot \left(1 + \frac{y_{\text{CO}_2}^{\text{out}}}{2}\right)}$ | $\frac{2}{2 - y_{\rm CO}^{\rm out}}$ | $\frac{1}{1 - y_{O_2}^{\text{out}}}$ | $\frac{\beta}{(1+\beta)\cdot y_s^{\text{out}}}$ |
| $y_{CO_2}^{out} =$ | $\frac{1-\chi}{1+\frac{\chi}{2}+\beta}$ | $\frac{1-(1+\beta)\cdot(2\cdot\alpha-2)}{\alpha\cdot(1+\beta)}$ | $\mathcal{Y}^{out}_{CO_2}$ | $\frac{2 - (3 + 2 \cdot \beta) \cdot y_{\text{CO}}^{\text{out}}}{2 \cdot (1 + \beta)}$ | $\frac{1 - (3 + 2 \cdot \beta) \cdot y_{0_2}^{\text{out}}}{1 + \beta}$ | $\frac{\left(1+2\cdot(1+\beta)\right)\cdot y_{s}^{\text{out}}-2\cdot\beta}{\beta}$ |
| $y_{\rm CO}^{\rm out} =$ | $\frac{\chi}{1+\frac{\chi}{2}+\beta}$ | $\frac{2 \cdot \alpha - 2}{\alpha}$ | $\frac{-(y_{\text{CO}_2}^{\text{out}}+2)\cdot(\beta\cdot y_{\text{CO}_2}^{\text{out}}+y_{\text{CO}_2}^{\text{out}}-1)}{\left(\frac{3}{2}+\beta\right)\cdot y_{\text{CO}_2}^{\text{out}}+3+2\cdot\beta}$ | y _{co} ^{out} | $2 \cdot y_{O_2}^{out}$ | $\frac{2 \cdot \beta - 2 \cdot (1 + \beta) \cdot y_s^{\text{out}}}{\beta}$ |
| $y_{0_2}^{out} =$ | $\frac{\frac{\chi}{2}}{1+\frac{\chi}{2}+\beta}$ | $\frac{\alpha-1}{\alpha}$ | $\frac{-(y_{CO_2}^{out}+2)\cdot(\beta\cdot y_{CO_2}^{out}+y_{CO_2}^{out}-1)}{(3+2\cdot\beta)\cdot y_{CO_2}^{out}+6+4\cdot\beta}$ | $\frac{y_{\rm CO}^{\rm out}}{2}$ | $y_{0_2}^{out}$ | $\frac{\beta - (1 + \beta) \cdot y_s^{\text{out}}}{\beta}$ |
| $y_s^{out} =$ | $\frac{\beta}{1+\frac{\chi}{2}+\beta}$ | $\frac{\beta}{(1+\beta)\cdot\alpha}$ | $\frac{\beta \cdot (2 + y_{\text{CO}_2}^{\text{out}})}{1 + 2 \cdot (1 + \beta)}$ | $\frac{\beta \cdot (2 - y_{\rm CO}^{\rm out})}{2 \cdot (1 + \beta)}$ | $\frac{\beta \cdot (1 - y_{0_2}^{\text{out}})}{1 + \beta}$ | y_s^{out} |

Table S.3. Overview of formulas for pure CO₂ conversion in the presence of a diluting agent, and how to obtain the conversion, flux ratio, or output fraction of CO₂, CO, O₂ or the diluting agent, when one of the other values is known. Note that β is the dilution ratio, as defined by Eq. (10) in the main paper.

S3. Formulas for the reverse Boudouard reaction

The reverse Boudouard reaction (RBR) is defined as follows:

$$CO_{2(g)} + C_{(s)} \rightleftharpoons 2 CO_{(g)}$$
(R.S1)

For this reaction, the standard reaction enthalpy ΔH° is equal to +172 kJ mol⁻¹, which is already 111 kJ mol⁻¹ lower compared to pure CO₂ splitting. For more general information on the plasma-based RBR, we refer to the available literature (e.g., Refs. [1-4]).

The volumetric gas flow rate again increases and needs to be taken into account. However, with respect to using the correct formulas, the main difference with CO_2 splitting is that the reactants are not all in gas phase. Furthermore, because the carbon material is usually in a fixed bed positioned after the plasma, there is no fraction of carbon at the inlet and $y_{C(s)}^{in}$ is zero.

Another difference is that O_2 will still be present in low amounts as by-product, originating from pure CO_2 splitting. Hence, there is some deviation from the "ideal" RBR reaction (R.S1). This means that the conversion cannot be determined solely based on the output CO_2 fraction $y_{CO_2}^{out}$, as was the case for pure CO_2 splitting. In case of the RBR, the fractions of all gaseous components need to be determined. Afterwards, the flux ratio can be derived through the O-balance equation, as all species containing O-atoms are quantified:

$$\alpha = \frac{2 \cdot y_{CO_2}^{in}}{2 \cdot y_{CO_2}^{out} + y_{CO}^{out} + 2 \cdot y_{O_2}^{out}}$$
(5.2)

With $y_{CO_2}^{ln}$ equal to 1 if no other gas is present at the inlet. Hence, the absolute CO₂ conversion, Obased selectivities and yields can already be determined in the same way as described in Sections 4.2 and 4.4 of the main paper.

To derive the consumption rate of solid carbon, we write down the carbon balance equation as:

$$\frac{\alpha \cdot (y_{\rm CO_2}^{\rm out} + y_{\rm CO}^{\rm out})}{y_{\rm CO_2}^{\rm in} + \frac{\dot{n}_{\rm C}_{\rm (s)}}{\dot{n}_{\rm CO_2}^{\rm in}}} = 1$$
(5.3)

With $\dot{n}_{C_{(s)}}$ equal to the consumption or loss rate of solid carbon from the fixed bed, expressed through Eq. (S.4):

$$\dot{n}_{C_{(S)}} = \dot{n}_{CO_2}^{in} \cdot \left(\alpha \cdot \left(y_{CO_2}^{out} + y_{CO}^{out} \right) - y_{CO_2}^{in} \right)$$
(S.4)

The advantage of this expression is that one can follow in real-time how much carbon is consumed and from which point the bed starts to get saturated. Weighing of the carbon material before and after reaction can still serve as validation.

The "fraction" of solid carbon consumed $y_{C_{(s)}}^{in}$ is then expressed as:

$$y_{C(s)}^{in} = \frac{\dot{n}_{C(s)}}{\dot{n}_{C(s)} + \dot{n}_{CO_2}^{in}}$$
(5.5)

Hence, the total conversion is defined through Eq. (S.6):

$$\chi^{\text{tot}} = y_{C_{(s)}}^{\text{in}} + \frac{\dot{n}_{\text{CO}_2}^{\text{in}}}{\dot{n}_{C_{(s)}} + \dot{n}_{\text{CO}_2}^{\text{in}}} \cdot \chi_{\text{CO}_2}^{\text{abs}}$$
(S.6)

The definitions for energy cost and energy efficiency remain the same as described in Section 4.3 of the main paper.

S4. Additional details for the comparison of energy efficiency with different equations, for DRM and CO₂ hydrogenation

The numerical example in Section 5.2 of the main paper describes the different energy efficiency equations applied to a dataset for DRM in a typical warm plasma, based on earlier experiments from our group [5]. An additional numerical example for a theoretical CO₂ hydrogenation dataset in a typical cold plasma (e.g., dielectric barrier discharge) is presented below.

Table S.4. Numerical example for CO_2 hydrogenation in a typical cold plasma with a SEI = 240.6 kJ mol⁻¹: flow rates of gases at the inlet, and unreacted gases and products at the outlet.

| Flow rate (mL min ⁻¹) | CO ₂ | H ₂ | CH₄ | со | H₂O | Total |
|--------------------------------------|-----------------|----------------|-----|-----|-----|-------|
| in | 15 | 25 | 0 | 0 | 0 | 40 |
| out | 13 | 21.5 | 0.5 | 1.5 | 2.5 | 39 |



Fig. S.1. Numerical example of the energy efficiency calculated according to the various definitions for CO_2 hydrogenation. The different equation numbers refer to the equations presented in the main paper. Eq. (52)* represents the chemical energy efficiency without taking all products into account, Eq. (52)** represents the chemical energy efficiency when the reaction enthalpy corresponding to the "ideal" stoichiometry is used (described by reaction (R3) for CO_2 methanation in the main paper).

The same conclusions from the main paper remain valid for this example. In fact, due to the overall lower energy efficiency, the relative differences between the formulas are even more striking. Also note that the chemical energy efficiency (Eq. (52)) is slightly negative in this case, confirming even more that this definition is not so convenient compared to the fuel energy efficiency equations (Eqs. (41), (49), (50) and (51)).

The energy efficiency values from Fig. 2 in the main paper and Fig. S.1 are listed in Table S.5. The lower heating values, formation enthalpies and reaction enthalpies used for the calculation of the energy efficiency values are listed in Table S.6.

Table S.5. Energy efficiency values, obtained through the various equations presented in the main paper, for both the DRM (see Fig. 2 from the main paper) and CO_2 hydrogenation (Fig. S.1) numerical example.

| Energy efficiency η (%) | y efficiency η (%) DRM (Fig. 2 main paper) | |
|-------------------------|---|------|
| Eq. (41) | 52.9 | 5.3 |
| Eq. (49) | 61.3 | 7.9 |
| Eq. (50) | 48.6 | 2.6 |
| Eq. (51) | 56.4 | 3.8 |
| Eq. (52) | 27.8 | -0.2 |
| Eq. (52)* | 38.9 | 1.5 |
| Eq. (52)** | 31.3 | -1.9 |

Table S.6. Overview of lower heating values [6], as well as the formation enthalpies & reaction enthalpies [7], used in the energy efficiency calculations.

| CO ₂ | 0 |
|-------------------------------|---|
| CH ₄ | 802 |
| СО | 283 |
| H ₂ | 242 |
| C ₂ H ₂ | 1255 |
| C ₂ H ₄ | 1324 |
| C_2H_6 | 1428 |
| H ₂ O | 0 |
| CO ₂ | -393.5 |
| CH₄ | -74.8 |
| СО | -110.5 |
| H ₂ | 0 |
| C_2H_2 | 226.7 |
| C_2H_4 | 52.4 |
| C ₂ H ₆ | -83.8 |
| H ₂ O | -241.8 |
| DRM | 247.3 |
| CO ₂ methanation | -164.9 |
| | $\begin{array}{c} {\rm CO}_2 \\ {\rm CH}_4 \\ {\rm CO} \\ {\rm H}_2 \\ {\rm C}_2 {\rm H}_2 \\ {\rm C}_2 {\rm H}_2 \\ {\rm C}_2 {\rm H}_4 \\ {\rm C}_2 {\rm H}_6 \\ {\rm H}_2 {\rm O} \\ {\rm CO}_2 \\ {\rm CH}_4 \\ {\rm CO} \\ {\rm CO} \\ {\rm H}_2 \\ {\rm C}_2 {\rm H}_4 \\ {\rm C}_2 {\rm H}_6 \\ {\rm H}_2 {\rm O} \\ {\rm DRM} \\ {\rm CO}_2 \text{ methanation} \end{array}$ |

S5. Derivation of flux ratio in the presence of a standard

S5.1. CO₂ splitting

When the standard component is present as part of the feed gas, the same equations can be used as the ones presented in Table S.3 in section S2. However, when the standard is added in the exhaust stream, the flux ratio at the end, α_s , is given by:

$$\alpha_s = \frac{\dot{n}_{\text{tot}}^{\text{out}} + \dot{n}_s}{\dot{n}_{\text{tot}}^{\text{in}}} = \frac{\alpha + \beta}{1} = 1 + \frac{\chi}{2} + \beta$$
(S.7)

Note that this is different from the equation in Table S.3, expressing the flux ratio as a function of conversion, due to the standard component not being present in the feed gas. As a result, all expressions for the flux ratio as a function of the conversion or one of the fractions, and vice versa, change compared to the ones presented in Table S.3. The equations to derive the flux ratio from each of the fractions are:

$$\alpha_{s} = 1 + \beta + \frac{1 - (1 + \beta) * y_{CO_{2}}^{out}}{2 + y_{CO_{2}}^{out}}$$
(S.8)

$$\alpha_s = (1+\beta) * \left(1 + \frac{y_{\rm CO}^{\rm out}}{2 - y_{\rm CO}^{\rm out}}\right)$$
(5.9)

$$\alpha_{s} = (1+\beta) * \left(1 + \frac{y_{0_{2}}^{\text{out}}}{1 - y_{0_{2}}^{\text{out}}}\right)$$
(5.10)

$$\alpha_s = \frac{\beta}{\gamma_s^{\text{out}}} \tag{S.11}$$

Note that the equations in Table S.3 to derive the conversion from one of the measured fractions and vice versa do not change, as they are independent from the position where the standard is added. Also, it should be kept in mind that the flux ratio before the standard component is added is still defined as $1 + \frac{\chi}{2}$. The relationship between the flux ratio before (α) and after (α_s) addition of the standard can be derived from Eq. (S.7):

$$\alpha_s - \beta = 1 + \frac{\chi}{2} = \alpha \tag{S.12}$$

S5.2. DRM & CO₂ hydrogenation

Fig. S.2 shows the different stages in the gas pathway for DRM or CO_2 hydrogenation with a cold trap and GC (or any other device for gas analysis):



Fig. S.2. Gas pathway for a DRM/CO₂ hydrogenation experiment with cold trap and GC. The green A and B represent the initial stages (i.e. during a blank measurement, when the plasma is off) of the gas mixture, before and after dilution of the feed gas by the standard, respectively. The purple C, D and E represent the stages of the gas mixture when the plasma is on: gas expansion or contraction due to the reaction stoichiometry (C), gas contraction through removal of liquid products (D) and again dilution of the gas stream by adding the standard (E).

During the different stages of the gas pathway, the total molar and volumetric flow rate changes several times, and along with the total flow rate also the fractions of the gaseous components. When expressing a flux ratio and a fraction of a certain component, it is important to use the corresponding values, that refer to the same situations. For example, when expressing the flux ratio between situation B (flow in) and E (flow out), both the measured input and output fractions can be used directly, and the flux ratio is equal to the ratio of the obtained fractions of the standard, during the blank and plasma measurement:

$$\alpha_s^{\text{meas}} = \frac{y_s^{\text{in}}}{y_s^{\text{out}}}$$
(S.13)

When expressing the flux ratio between situation A (flow in) and E (flow out), the measured output fractions can still be used directly, but the measured input factors not, since they do not represent the situation at the reactor inlet where no standard is present. To account for this, the factor $(1 + \beta)$ is multiplied with the measured input fractions, as well as the flux ratio:

$$y_i^{\text{in}} = y_i^{\text{in,meas}} \cdot (1 + \beta) \tag{(5.14)}$$

$$\alpha_s^{\text{fin}} = \alpha_s^{\text{meas}} \cdot (1 + \beta) \tag{S.15}$$

With β still representing the dilution ratio, generally defined as:

$$\beta = \frac{\dot{n}_s}{\dot{n}_{\text{tot}}^{\text{in}}} \tag{S.16}$$

With \dot{n}_s the molar flow rate of the standard s and \dot{n}_{tot}^{in} the total molar flow rate at the reactor inlet. Indeed, expressing Eq. (S.15) in terms of molar flow rates becomes:

$$\alpha_s^{\text{fin}} = \frac{\dot{n}_{\text{tot}}^{\text{out}} + \dot{n}_s}{\dot{n}_{\text{tot}}^{\text{in}} + \dot{n}_s} \cdot \left(1 + \frac{\dot{n}_s}{\dot{n}_{\text{tot}}^{\text{in}}}\right)$$
(S.17)

With \dot{n}_{tot}^{out} the total molar flow rate before addition of the standard, but after removal of liquid components (stage D in Fig. S.2). After rearrangement this results in:

$$\alpha_s^{\text{fin}} = \frac{\dot{n}_{\text{tot}}^{\text{out}} + \dot{n}_s}{\dot{n}_{\text{tot}}^{\text{in}}} \tag{S.18}$$

This proves that Eq. (S.15) represents the ratio of the flow rate in stage E relative to the flow rate in stage A. A similar derivation can be performed for Eq. (S.14) expressing the input fractions.

Note that in the main paper, we defined an "initial" flux ratio (after reaction, but before condensation/deposition of certain products) and a "final" flux ratio (after condensation/deposition of certain products) in the main paper. In this context, this corresponds with the flux ratio between stage A and C, and between A and D, respectively. The relationship between the two was given by Eq. (43) in the main paper:

$$\alpha^{\text{init}} = \frac{\alpha^{\text{fin}}}{(1 - \sum_{k} y_{k}^{\text{out}})} \tag{S.19}$$

In order to determine the initial flux ratio, the relationship between the flux ratio after condensation, but before the standard is added (α^{fin}) and the flux ratio after the standard is added (α^{fin}_{s}) has to be derived.

Consider again Eq. (S.17). When we divide every value by the total molar flow rate at the reactor inlet, we obtain:

$$\alpha_{s}^{\text{fin}} = \frac{\frac{\dot{n}_{\text{tot}}^{\text{out}}}{\dot{n}_{\text{tot}}^{\text{in}}} + \frac{\dot{n}_{s}}{\dot{n}_{\text{tot}}^{\text{in}}} + \frac{\dot{n}_{s}}{\dot{n}_{\text{tot}}^{\text{in}}} \cdot \left(\frac{\dot{n}_{\text{tot}}^{\text{out}}}{\dot{n}_{\text{tot}}^{\text{in}}} + \frac{\dot{n}_{s}}{\dot{n}_{\text{tot}}^{\text{in}}}\right)}{\frac{\dot{n}_{\text{tot}}^{\text{in}}}{\dot{n}_{\text{tot}}^{\text{in}}} + \frac{\dot{n}_{s}}{\dot{n}_{\text{tot}}^{\text{in}}}}$$
(S.20)

We can rewrite this as:

$$\alpha_{s}^{\text{fin}} = \frac{\alpha^{\text{fin}} + \beta + \beta \cdot (\alpha^{\text{fin}} + \beta)}{1 + \beta}$$
(S.21)

Note that the ratio of the total outflow over the total inflow is indeed equal to α^{fin} . This results in:

$$\alpha_{\rm s}^{\rm fin} = \alpha^{\rm fin} + \beta \tag{S.22}$$

All expressions for the flux ratios and fractions are summarized in Table S.7. Note that a single row gives a flux ratio and an in- and output fraction that needs to be used together, when calculating each performance metric. In other words, when one uses for example the measured in- and output fractions to calculate the absolute conversion, the corresponding flux ratio (Eq. (S.13)) should be used. The only exception is the calculation of the effective conversion, where always the fraction at the reactor inlet (Eq. (S.14), corresponding to stage A in Fig. S.2) has to be used, even when the measured fractions and flux ratio are used to calculate the absolute conversion:

$$\chi_i^{\text{eff}} = y_i^{\text{in,meas}} \cdot (1+\beta) \cdot \chi_i^{\text{abs}} = y_i^{\text{in,meas}} \cdot (1+\beta) \cdot \frac{y_i^{\text{in,meas}} - \alpha_s^{\text{meas}} \cdot y_i^{\text{out,meas}}}{y_i^{\text{in,meas}}}$$
(S.23)

Using only the diluted fraction of the reactants $y_i^{\text{in,meas}}$ without the factor $(1 + \beta)$ will lead to an underestimation of the effective conversion, and hence also an underestimation of the total conversion, an overestimated energy cost, etc.

| Stages | Flux ratio | Input fraction $y_i^{in} =$ | Output fraction $y_{i/j}^{out} =$ |
|-----------------|--|--|---|
| ln: B Out: E | $\alpha_s^{\text{meas}} = \frac{y_s^{\text{in}}}{y_s^{\text{out}}}$ | $y_i^{ m in,meas}$ | $y_{i/j}^{\text{out,meas}}$ |
| In: A Out: E | $\alpha_s^{\rm fin} = \alpha_s^{\rm meas} \cdot (1+\beta)$ | $y_i^{\text{in,meas}} \cdot (1 + \beta)$ | $y_{i/j}^{\text{out,meas}}$ |
| In: A Out: D | $\alpha^{\rm fin} = \alpha^{\rm fin}_s - \beta$ | $y_i^{\text{in,meas}} \cdot (1 + \beta)$ | $y_{i/j}^{\text{out,meas}} \cdot \left(1 + \frac{\beta}{\alpha^{\text{fin}}}\right)$ |
| In: A Out: C | $\alpha^{\text{init}} = \frac{\alpha^{\text{fin}}}{1 - \sum_k y_k^{\text{out}}}$ | $y_i^{\text{in,meas}} \cdot (1 + \beta)$ | $y_{i/j}^{\text{out,meas}} \cdot \left(1 + \frac{\beta}{\alpha^{\text{fin}}}\right) \cdot \left(1 - \sum_{k} y_{k}^{\text{out}}\right)$ |

Table S.7. Expressions for flux ratio, input fraction and output fraction at each stage of the gas pathway for DRM/CO₂ hydrogenation. The parameters on the same row should be used together to calculate the performance metrics.

S6. Experimental details for pure CO₂ splitting in DBD plasma

Two separate series of experiments were performed (one with a standard component, one with a flow meter) with CO_2 (Air Liquide, purity 99.999 %) flowing in a DBD plasma reactor with an input flow rate of 100 mL_s min⁻¹. When using a standard, an input flow rate of 20 mL_s min⁻¹ N₂ was used. An Agilent 990 Micro GC was calibrated for CO_2 , CO, O_2 and N₂ to analyze the fractions in the output gas mixture. The flow rate was regulated by Bronkhorst mass flow controllers, which were calibrated manually with a Sensidyne Gilibrator-3 calibrator, before the GC calibration and experiments.

For each experimental run, three samples of the input gas mixture were injected on the GC, with 5 minutes between each sample. At least eleven samples were taken of the output gas mixture, of which at least seven after reaching a steady-state. From the latter, the average of the obtained peak areas was used for further calculations. When the bubble flow meter was used, two flow rate measurements were performed for each GC sample taken. After performing these experimental runs in triplicate, the weighted averages of the performance metrics were taken. For the errors on each individual measurement, we consider both the errors on the GC calibration factors, and the standard deviation on the obtained peak areas. The plasma power was monitored and confirmed to remain constant (approximately 60 W) during the different experiments.

The output fractions of CO₂, CO, O₂ and the standard (if present), as well as the dilution factor β (obtained through the MFC calibration), the measured plasma power and specific energy input (SEI) are presented in Table S.8. Each value is a weighted average of three separate experiments. The weighted averages of the flux ratio('s) and performance metrics, calculated through the different methods, are presented in Table S.9 for the experiments with the standard, and in Table S.10 for the experiments with the flow meter.

| | Standard | Flow meter |
|------------------------------------|----------------------------|-------------|
| $y_{CO_2}^{out}(\%)$ | 73.8 ± 0.3 | 88.9 ± 0.4 |
| y _{co} ^{out} (%) | 6.25 ± 0.02 7.34 ± 0.02 | |
| $y_{0_2}^{out}(\%)$ | 3.17 ± 0.02 | 3.58 ± 0.02 |
| y _s ^{out} (%) |) 16.8 ± 0.2 / | |
| β | 0.211 ± 0.001 | / |
| P (W) | 7) 59.8 ± 0.1 59.93 ± 0.06 | |
| SEI (kJ/mol) | 832 ± 2 | 808 ± 7 |

Table S.8. Weighted averages of all measured values and the SEI (kJ/mol), for both the experiments with the standard and the flow meter.

| Known → Unknown ↓ | $y_{CO_2}^{out}$ | y _{co} ^{out} | $y_{0_2}^{out}$ | $y_s^{ m out}$ |
|----------------------|------------------------------|--------------------------------|--------------------------------|---------------------------|
| χ (%) = | 7.8 ± 0.3 | 7.81 ± 0.02 | 7.93 ± 0.05 | 8 ± 3 |
| $\alpha_s =$ | 1.250 ± 0.002 | 1.2496 ± 0.0006 | 1.2502 ± 0.0007 | 1.25 ± 0.02 |
| α = | 1.039 ± 0.002 | 1.0391 ± 0.0009 | 1.0397 ± 0.0009 | 1.04 ± 0.02 |
| <i>EC</i> (kJ/mol) = | (10.6 ± 0.4)·10 ³ | (10.65 ± 0.04)·10 ³ | (10.49 ± 0.07)·10 ³ | $(10 \pm 4) \cdot 10^3$ |
| EY(mol/kJ) = | (9.4 ± 0.4)·10 ⁻⁵ | (9.39 ± 0.03)·10 ⁻⁵ | (9.53 ± 0.07)·10 ⁻⁵ | (10 ± 4)·10 ⁻⁵ |
| η (%) = | 2.7 ± 0.1 | 2.66 ± 0.01 | 2.70 ± 0.02 | 3±1 |

Table S.9. Weighted averages of the conversion, flux ratio's, energy cost, energy yield and energy efficiency, calculated from each output fraction (incl. the standard) for the experiments performed with a standard component added.

Table S.10. Weighted averages of the conversion, flux ratio, energy cost, energy yield and energy efficiency, calculated from each measured output fraction and the measured flux ratio for the experiments performed with a flow meter.

| Known → Unknown ↓ | $y_{\rm CO_2}^{\rm out}$ | $y_{\rm CO}^{\rm out}$ | $y_{0_2}^{out}$ | α |
|----------------------|------------------------------|--------------------------------|------------------------------|-------------------------|
| χ (%) = | 7.6 ± 0.3 | 7.61 ± 0.02 | 7.42 ± 0.05 | 7 ± 2 |
| α = | 1.038 ± 0.001 | 1.0381 ± 0.0001 | 1.0371 ± 0.0002 | 1.04 ± 0.01 |
| <i>EC</i> (kJ/mol) = | (10.6 ± 0.4)·10 ³ | $(10.62 \pm 0.09) \cdot 10^3$ | $(10.9 \pm 0.1) \cdot 10^3$ | $(11 \pm 4) \cdot 10^3$ |
| EY(mol/kJ) = | (9.5 ± 0.3)·10 ⁻⁵ | (9.42 ± 0.08)·10 ⁻⁵ | (9.2 ± 0.1)·10 ⁻⁵ | (9 ± 3)·10⁻⁵ |
| η (%) = | 2.68 ± 0.09 | 2.67 ± 0.02 | 2.60 ± 0.03 | 2.5 ± 0.8 |

S7. Experimental details for DRM in the presence of O₂, in APGD plasma

For this second example, we look at CO₂ conversion in the presence of CH₄ and O₂, specifically in a 42.5/42.5/15 CO₂/CH₄/O₂ ratio. This reaction is also called oxidative CO₂ reforming of methane (OCRM). Again, two separate series of experiments were performed (one with a standard, one with a flow meter). The gases were again provided by gas bottles from Air Liquide (purity 99.999 %), and were sent to an APGD plasma reactor through Bronkhorst MFCs, which were calibrated manually with the bubble flow meter. An Agilent 990 Micro GC was calibrated for CO₂, CH₄, CO, H₂, C₂H₂, C₂H₄, C₂H₆, O₂ and N₂ to analyze the fractions in the output gas mixture. A sample of the input gas mixture was sent three times onto the GC, with 5 minutes between each sample. The same procedure was done for the output gas mixture, after the plasma reached a steady-state. The average of the obtained peak areas was used for further calculations. Each experimental run was repeated three times, and the weighted average of the performance metrics was taken. For the errors on each individual measurement, we consider both the errors on the GC calibration factors, and the standard deviation on the obtained peak areas.

The total reactant input flow rate with the standard method was $1 L_n min^{-1}$, and the standard was again N_2 , with a flow rate of $100 mL_n min^{-1}$ added to the effluent gas mixture between the cold trap and the GC sample point. For the experiments that were performed with the bubble flow meter, the flow rate was changed to $1 L_s min^{-1}$. The plasma power was monitored through the readings on a DC Technix power supply, and confirmed to remain constant (91-94 W) during the different experiments.

The different fractions for each component (incl. the standard, if present), as well as the dilution ratio β (obtained through the MFC calibration), the measured plasma power and specific energy input (SEI) are presented in Table S.11 for both the experiments with the standard (left) and the flow meter (right). Each value is a weighted average of three separate experiments. The weighted averages of the flux ratios, atom balances and performance metrics, calculated through the different methods, are presented in Table S.12 for both the experiments with the standard (left) and the flow meter (right).

| | Standard | Flow meter |
|-----------------------------------|--------------|--------------|
| $y_{CO_2}^{in}(\%)$ | 38.4 ± 0.2 | 43.0 ± 0.2 |
| $y_{CH_4}^{in}(\%)$ | 38.3 ± 0.1 | 42.2 ± 0.2 |
| y ⁱⁿ ₀₂ (%) | 14.41 ± 0.03 | 16.05 ± 0.03 |
| $y_s^{in}(\%)$ | 9.37 ± 0.02 | / |
| $y_{CO_2}^{out}(\%)$ | 17.7 ± 0.2 | 17.5 ± 0.2 |

Table S.11. Weighted averages of all measured values and the SEI (kJ/mol), for both the experiments with the standard and the flow meter.

| $y_{CH_4}^{out}(\%)$ | 8.9 ± 0.2 | 8.7 ± 0.3 |
|------------------------------------|-----------------|-----------------|
| $y_{0_2}^{out}(\%)$ | 1.6 ± 0.1 | 1.5 ± 0.1 |
| y _{co} ^{out} (%) | 34.3 ± 0.3 | 40.0 ± 0.3 |
| $y_{H_2}^{out}(\%)$ | 28.5 ± 0.2 | 31.5 ± 0.3 |
| $y_{C_2H_2}^{out}(\%)$ | 1.31 ± 0.02 | 1.41 ± 0.03 |
| $y^{\text{out}}_{C_2H_4}(\%)$ | 0.376 ± 0.008 | 0.389 ± 0.007 |
| $\mathcal{Y}^{out}_{C_2H_6}(\%)$ | 0.0929 ± 0.0006 | 0.0800 ± 0.0006 |
| y _s ^{out} (%) | 7.57 ± 0.07 | / |
| β | 0.0977 ± 0.0004 | / |
| P (W) | 91.3 ± 0.9 | 94 ± 1 |
| <i>SEI</i> (kJ/mol) | 117 ± 1 | 127 ± 2 |

Table S.12. Weighted averages of all calculated values based on the different methods, for both the experiments with the standard and the flow meter.

| | Stoichiometric | Standard | Stoichiometric | Flow meter |
|---------------------------|-------------------|-----------------|-----------------|-----------------|
| $\alpha_s^{\rm meas}$ | 1.102 ± 0.008 | 1.24 ± 0.01 | / | / |
| α_s^{fin} | 1.307 ± 0.009 | 1.36 ± 0.01 | / | / |
| $lpha^{	ext{fin}}$ | 1.209 ± 0.009 | 1.26 ± 0.01 | 1.23 ± 0.01 | 1.24 ± 0.01 |
| $lpha^{	ext{init}}$ | 1.42 ± 0.01 | 1.43 ± 0.02 | 1.45 ± 0.01 | 1.45 ± 0.02 |

| $\chi^{abs}_{CO_2}(\%)$ | 45.2 ± 0.8 | 43.1 ± 0.8 | 49.9 ± 0.8 | 49.0 ± 0.9 |
|---|---------------|-------------|---------------|-------------|
| $\chi^{abs}_{CH_4}(\%)$ | 72.3 ± 0.6 | 71.3 ± 0.6 | 74.9 ± 0.8 | 74.6 ± 0.9 |
| $\chi^{abs}_{0_2}(\%)$ | 87 ± 1 | 86 ± 1 | 88.3 ± 0.9 | 88.1 ± 0.9 |
| $\chi^{\rm eff}_{\rm CO_2}(\%)$ | 19.1 ± 0.4 | 18.2 ± 0.4 | 21.4 ± 0.4 | 21.1 ± 0.4 |
| $\chi^{\rm eff}_{\rm CH_4}(\%)$ | 30.4 ± 0.3 | 29.9 ± 0.3 | 31.6 ± 0.4 | 31.5 ± 0.4 |
| $\chi^{\mathrm{eff}}_{\mathrm{0}_{2}}(\%)$ | 13.7 ± 0.2 | 13.6 ± 0.2 | 14.2 ± 0.2 | 14.1 ± 0.2 |
| $\chi^{tot}(\%)$ | 63.1 ± 0.5 | 61.7 ± 0.5 | 67.4 ± 0.6 | 66.9 ± 0.6 |
| S ^C _{CO} (%) | 91±2 | 97 ± 2 | 92 ± 2 | 93 ± 2 |
| $S^{C}_{C_2H_2}(\%)$ | 6.9 ± 0.2 | 7.4 ± 0.2 | 6.4 ± 0.2 | 6.6 ± 0.2 |
| $S_{C_2H_4}^{C}(\%)$ | 1.99 ± 0.05 | 2.12 ± 0.06 | 1.82 ± 0.05 | 1.86 ± 0.06 |
| S ^C _{C2H6} (%) | 0.491 ± 0.009 | 0.52 ± 0.01 | 0.365 ± 0.007 | 0.38 ± 0.01 |
| S ^H _{H2} (%) | 61±1 | 65 ± 1 | 61±1 | 62 ± 1 |
| $S^{\mathrm{H}}_{C_{2}\mathrm{H}_{2}}(\%)$ | 2.82 ± 0.06 | 2.97 ± 0.06 | 2.70 ± 0.07 | 2.75 ± 0.07 |
| $S^{\mathrm{H}}_{\mathrm{C}_{2}\mathrm{H}_{4}}(\%)$ | 1.63 ± 0.04 | 1.71 ± 0.04 | 1.51 ± 0.04 | 1.54 ± 0.04 |
| $S^{\mathrm{H}}_{\mathrm{C_2H_6}}(\%)$ | 0.600 ± 0.009 | 0.63 ± 0.01 | 0.459 ± 0.009 | 0.47 ± 0.01 |
| $S_{\mathrm{H_20}}^{\mathrm{H}}(\%)$ | 34.3 ± 0.05 | 28.7 ± 0.5 | 35 ± 2 | 33.4 ± 0.7 |

| S ⁰ _{C0} (%) | 68 ± 1 | 73 ± 1 | 68 ± 1 | 70 ± 1 |
|--|---------------|---------------|---------------|---------------|
| S ⁰ _{H20} (%) | 31.7 ± 0.4 | 26.9 ± 0.4 | 32 ± 2 | 30.1 ± 0.6 |
| <i>Y</i> ^C _{C0} (%) | 53.1±0.6 | 55.0 ± 0.7 | 57.4 ± 0.7 | 57.9 ± 0.9 |
| $Y_{C_2H_2}^{C}(\%)$ | 4.05 ± 0.07 | 4.21 ± 0.07 | 4.03 ± 0.09 | 4.1 ± 0.1 |
| $Y_{C_2H_4}^{C}(\%)$ | 1.17 ± 0.03 | 1.21 ± 0.03 | 1.12 ± 0.02 | 1.13 ± 0.03 |
| $Y_{C_2H_6}^{C}(\%)$ | 0.288 ± 0.003 | 0.299 ± 0.004 | 0.231 ± 0.003 | 0.235 ± 0.004 |
| <i>Y</i> ^H _{H2} (%) | 44.3 ± 0.5 | 45.9 ± 0.5 | 45.9 ± 0.6 | 46.5 ± 0.7 |
| $Y_{C_2H_2}^{H}(\%)$ | 2.03 ± 0.04 | 2.11 ± 0.04 | 2.03 ± 0.05 | 2.06 ± 0.05 |
| $Y^{\rm H}_{\rm C_2H_4}(\%)$ | 1.17 ± 0.03 | 1.21 ± 0.03 | 1.13 ± 0.02 | 1.14 ± 0.03 |
| $Y^{\rm H}_{\rm C_2H_6}(\%)$ | 0.433 ± 0.005 | 0.449 ± 0.005 | 0.350 ± 0.005 | 0.356 ± 0.006 |
| $Y_{\rm H_{2}0}^{\rm H}(\%)$ | 24.7 ± 0.3 | 20.0 ± 0.3 | 27 ± 2 | 25.1 ± 0.5 |
| Y ⁰ _{C0} (%) | 38.6 ± 0.5 | 40.0 ± 0.5 | 41.4 ± 0.5 | 41.8 ± 0.6 |
| Y ⁰ _{H20} (%) | 17.9 ± 0.2 | 14.5 ± 0.2 | 19±1 | 17.9 ± 0.3 |
| b _c | 1.00 ± 0.01 | 1.04 ± 0.01 | 1.00 ± 0.01 | 1.01 ± 0.01 |
| b _H | 1.01 ± 0.01 | 0.99 ± 0.01 | 1.01 ± 0.02 | 1.00 ± 0.01 |
| b ₀ | 1.00 ± 0.01 | 1.00 ± 0.01 | 1.00 ± 0.02 | 1.00 ± 0.01 |

| EC (kJ/mol) | 179 ± 3 | 183 ± 3 | 186 ± 3 | 188 ± 3 |
|---------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|
| EY (mol/kJ) | (5.60 ± 0.08)·10 ⁻³ | (5.47 ± 0.08)·10 ⁻³ | (5.37 ± 0.09)·10 ⁻³ | (5.33 ± 0.09)·10 ⁻³ |
| η (%) (Eq. (41)) | 54.8 ± 0.6 | 56.7 ± 0.6 | 56.2 ± 0.6 | 56.9 ± 0.8 |
| η (%) (Eq. (49)) | 69.4 ± 0.9 | 73 ± 1 | 68±1 | 70 ± 1 |
| η (%) (Eq. (50)) | 48.2 ± 0.5 | 49.9 ± 0.5 | 50.0 ± 0.6 | 50.6 ± 0.7 |
| η (%) (Eq. (51)) | 61.0 ± 0.8 | 63.9 ± 0.9 | 60.7 ± 0.9 | 62 ± 1 |
| η (%) (Eq. (52)) | 2 ± 1 | 5 ± 1 | 3 ± 3 | 3 ± 2 |

S8. References

[1] F. Girard-Sahun, O. Biondo, G. Trenchev, G.J. van Rooij, A. Bogaerts, Carbon bed post-plasma to enhance the CO2 conversion and remove O2 from the product stream, Chemical Engineering Journal 442 (2022) 136268. <u>https://doi.org/10.1016/j.cej.2022.136268</u>.

[2] J. Huang, H. Zhang, Q. Tan, L. Li, R. Xu, Z. Xu, X. Li, Enhanced conversion of CO2 into O2-free fuel gas via the Boudouard reaction with biochar in an atmospheric plasmatron, Journal of CO2 Utilization 45 (2021). <u>https://doi.org/10.1016/j.jcou.2020.101429</u>.

[3] Y. Wu, S.-Z. Li, Y.-L. Niu, H. Yan, D. Yang, J. Zhang, Experimental investigation of CO2 conversion in Boudouard reaction driven by an atmospheric-pressure microwave plasma torch, Journal of Physics D: Applied Physics (2022). <u>https://doi.org/10.1088/1361-6463/acaeda</u>.

[4] H. Zhang, Q. Tan, Q. Huang, K. Wang, X. Tu, X. Zhao, C. Wu, J. Yan, X. Li, Boosting the Conversion of CO2 with Biochar to Clean CO in an Atmospheric Plasmatron: A Synergy of Plasma Chemistry and Thermochemistry, ACS Sustainable Chemistry & Engineering 10(23) (2022) 7712-7725. https://doi.org/10.1021/acssuschemeng.2c01778.

[5] B. Wanten, S. Maerivoet, C. Vantomme, J. Slaets, G. Trenchev, A. Bogaerts, Dry reforming of methane in an atmospheric pressure glow discharge: Confining the plasma to expand the performance, Journal of CO2 Utilization 56 (2022) 101869.

https://doi.org/10.1016/j.jcou.2021.101869.

[6] T.E. Toolbox, Gases - Gross and Net Heat Values, 2003.

https://www.engineeringtoolbox.com/gross-net-heating-values-d_420.html. (Accessed 25 April 2023).

[7] B. Szyk, D. Czernia, Enthalpy Calculator. <u>https://www.omnicalculator.com/physics/enthalpy</u>. (Accessed 25 April 2023).

List of symbols

| Symbols (incl. sub- and superscripts) | Description |
|---------------------------------------|---|
| * | Incorrect expression |
| A | Atom |
| abs | Absolute |
| app | Applied |
| α | Flux ratio |
| b | Balance |
| β | Dilution ratio |
| chem | Chemical |
| conv | Converted |
| corr | Corrected |
| des | Desired |
| DF | Dissociation fraction |
| ΔH° | Standard reaction enthalpy (kJ mol ⁻¹) |
| E | Energy (kJ ⁻¹ , eV) |
| EC | Energy cost (kJ mol ⁻¹ , kJ L ⁻¹ , eV molecule ⁻¹) |
| eff | Effective |
| EY | Energy yield (mol kJ ⁻¹ , L kJ ⁻¹ , molecule eV ⁻¹) |
| fin | Final |
| H _f | Formation enthalpy (kJ mol ⁻¹) |
| i | Reactant |
| in | At reactor inlet/before reactor |
| init | Initial |
| j | (Gaseous) product |
| k | Condensed/deposited product |
| LHV | Lower heating value (kJ mol ⁻¹) |
| meas | Measured |
| μ | Stoichiometric coefficient |
| <u> </u> | Molar flow (mol min ⁻¹) |
| N _A | Avogadro's constant (molecule mol ⁻¹) |
| η | Energy efficiency |
| out | At reactor outlet/after reactor |
| <i>p</i> | Pressure (bar, atm) |
| P | Power (kW) |
| RE | Relative error |
| reac | Reaction |
| rep | Reported |
| <u> </u> | Diluting agent/standard component |
| <i>S</i> | Selectivity |
| SEI | Specific energy input (kJ mol ⁻¹ , kJ L ⁻¹ , eV |
| | molecule ⁻¹) |
| <u>T</u> | remperature (K, °C) |
| tot | Iotal |

| Ϋ́ | Volume flow (L min ⁻¹) |
|----------------|-------------------------------------|
| V _m | Molar volume (L mol ⁻¹) |
| X | Conversion |
| y | Molar/volume fraction |
| Y | Product yield |